

LEGISLATIVE RESEARCH COMMISSION

Investment Rates of Return, Governance, and Policies of the Kentucky Retirement Systems and the Kentucky Teachers' Retirement System

Research Report No. 352

Prepared by

Rick Graycarek, Margaret Hurst, and Colleen Kennedy

Kentucky Legislative Research Commission

David L. Williams President, LRC Co-Chair

SENATE

Katie Kratz Stine **President Pro Tem**

Dan Kelly Majority Floor Leader

Ed Worley Minority Floor Leader

Dan Seum Majority Caucus Chair

Johnny Ray Turner Minority Caucus Chair

Carroll Gibson Majority Whip

Jerry P. Rhoads Minority Whip Gregory D. Stumbo Speaker, LRC Co-Chair

HOUSE

Larry Clark Speaker Pro Tem

Rocky Adkins Majority Floor Leader

Jeff Hoover Minority Floor Leader

Robert R. Damron Majority Caucus Chair

Bob DeWeese Minority Caucus Chair

> John Will Stacy Majority Whip

> David Floyd Minority Whip

Robert Sherman, Director

The Kentucky Legislative Research Commission is a 16-member committee comprised of the majority and minority leadership of the Kentucky Senate and House of Representatives. Under Chapter 7 of the Kentucky Revised Statutes, the Commission constitutes the administrative office for the Kentucky General Assembly. Its director serves as chief administrative officer of the legislature when it is not in session. The Commission and its staff, by law and by practice, perform numerous fact-finding and service functions for members of the General Assembly. The Commission provides professional, clerical, and other employees required by legislators when the General Assembly is in session and during the interim period between sessions. These employees, in turn, assist committees and individual members in preparing legislation. Other services include conducting studies and investigations, organizing and staffing committee meetings and public hearings, maintaining official legislative records and other reference materials, furnishing information about the legislature to the public, compiling and publishing administrative regulations, administering a legislative intern program, conducting a presession orientation conference for legislators, and publishing a daily index of legislative activity during sessions of the General Assembly.

The Commission also is responsible for statute revision; publication and distribution of the *Acts* and *Journals* following sessions of the General Assembly; and maintenance of furnishings, equipment, and supplies for the legislature.

The Commission functions as Kentucky's Commission on Interstate Cooperation in carrying out the program of the Council of State Governments as it relates to Kentucky.

Investment Rates of Return, Governance, and Policies of the Kentucky Retirement Systems and the Kentucky Teachers' Retirement System

Program Review and Investigations Committee

Sen. Ernie Harris, Co-chair Rep. Reginald Meeks, Co-chair

Sen. Charlie Borders Sen. Vernie McGaha Sen. R.J. Palmer Sen. Joey Pendleton Sen. Dan Seum Sen. Brandon Smith Sen. Katie Stine Rep. Dwight D. Butler Rep. Leslie Combs Rep. Rick Nelson Rep. Ruth Ann Palumbo Rep. Rick Rand Rep. Arnold Simpson Rep. Ken Upchurch

Greg Hager, Ph.D. Committee Staff Administrator

Project Staff

Rick Graycarek Margaret Hurst Colleen Kennedy

Research Report No. 352

Legislative Research Commission

Frankfort, Kentucky lrc.ky.gov

Adopted June 12, 2008

Paid for with state funds. Available in alternative form by request.

Reports Adopted by the Program Review and Investigations Committee, 1996 to 2008

Investment Rates of Return, Governance, and Policies of the Kentucky Retirement Systems and the Kentucky Teachers' Retirement System, Report 352, 2008

Medicaid Prescription Drug Benefit Fraud, Abuse, and Cost Management, 2007

Kentucky Housing Corporation, 2007

Siting of Electric Transmission Lines, Report 348, 2007

School Insurance, Report 347, 2007

Drug Courts, Report 346, 2007

Pollution Cap and Trade Programs in Kentucky, Report 343, 2007

Kentucky's Foster Care Program Is Improving, but Challenges Remain, Report 342, 2006

Planning for School Facilities Can Be Improved To Better Serve the Needs of All Students, Report 341, 2006

Kentucky's Community Mental Health System Is Expanding and Would Benefit From Better Planning and Reporting, Report 340, 2006

Highly Skilled Educator Program, Report 339, 2006

School Size and Student Outcomes in Kentucky's Public Schools, Report 334, 2006

Information Systems Can Help Prevent, but Not Eliminate, Health Care Fraud and Abuse, Report 333, 2006

Implications of the U.S. Supreme Court's *Kelo* Decision for the Use of Eminent Domain in Kentucky, Report 330, 2005

Planning for Water Projects in Kentucky: Implementation of Senate Bill 409, Report 329, 2005

Kentucky Can Improve the Coordination of Protective Services for Elderly and Other Vulnerable Adults, Report 327, 2005

Improved Coordination and Information Could Reduce the Backlog of Unserved Warrants, Report 326, 2005.

Offshore Outsourcing of Kentucky State Government Services: Direct Contracting Is Limited but the Amount of Subcontracting Is Unknown, Report 325, 2005

Appropriate Management and Technology Can Reduce Costs and Risks of Computer Use by State Employees, Report 324, 2004

Uncollected Revenues and Improper Payments Cost Kentucky Millions of Dollars a Year, Report 322, 2004

Improving Fiscal Accountability and Effectiveness of Services in the Kentucky Transitional Assistance Program, Report 321, 2004 Human Service Transportation Delivery, Report 319, 2004

The Commonwealth Accountability Testing System, Report 312, 2003

Postsecondary Education in Kentucky: Systemwide Improvement but Accountability Is Insufficient, Report 311, 2003

The SEEK Formula for Funding Kentucky's School Districts: An Evaluation of Data, Procedures, and Budgeting, Report 310, 2002

East and West Kentucky Corporations, Report 308, 2002

An Analysis of Kentucky's Prevailing Wage Laws and Procedures, Report 304, 2001

Executive Branch Contracting for Services: Inconsistent Procedures Limit Accountability and Efficiency, Report 303, 2001

Performance-based Budgeting: Concepts and Examples, Report 302, 2001

Impact Plus, Report 300, 2001

Kentucky Housing Corporation Allocation of Federal Homeless Grant Money, Report 291, 2000

Progress Report on Coordinated Human Service Transportation System, Report 298, 1999

Personnel Pilot Projects: Design Weakness Limits Effectiveness, Report 295, 1999

Kentucky Early Intervention System - First Steps, Report 293, 1999

Health Insurance Market for Employees and Retirees of Kentucky State Government, Report 286, 1999

State Agency Service Contract Administration, Report 285, 1999

Review of the Kentucky Children's Health Insurance Program, Report 283, 1999

Motor Vehicle Registration Abuse, Report 282, 1999

Kentucky Medicaid Drug File and Prior Authorization System, Report 281, 1999

Division of Licensing and Regulation, Cabinet for Health Services, Office of Inspector General, Report 279, 1997

State Park Marinas, Report 278, 1997

Department for Social Insurance Eligibility Determination Process, Report 277, 1996

Cabinet for Human Resources Family Service Workers' Caseloads, Report 275, 1996

Kentucky Medical Assistance Program, Report 274, 1996

Foreword

Many individuals provided valuable assistance. The authors want to thank the staffs of the Kentucky Retirement Systems and the Kentucky Teachers' Retirement Systems for their cooperation. Particular appreciation is extended to Mike Burnside, executive director of the Kentucky Retirement Systems; and Gary Harbin, executive secretary of the Kentucky Teachers' Retirement System.

Special appreciation is extended to staff members of the Kentucky League of Cities, the Kentucky Education Association, the Kentucky Retired Teachers' Association, and the Kentucky Transportation Employees' Association for their willingness to meet and share their thoughts and experiences.

The authors also want to thank Keith Brainard of the National Association of State Retirement Administrators and Kelly Haverstick of the Center for Retirement Research at Boston College for taking time to discuss retirement investment issues.

> Robert L. Sherman Director

Legislative Research Commission Frankfort, Kentucky June 12, 2008

Contents

Summary		.ix
Chapter 1:	Overview and Background	1
	Description of This Study	
	How This Study Was Conducted	
	Organization of the Report	
	Major Conclusions	
	Basic Finance and Investment Principles	
	Trade-off Between Risk and Return.	
	Investment Is Long Term	
	Diversification Reduces Risk	
	Liquidity	
	Employee Retirement Income Security Act	
	National Perspective	
	Governance	
	General Characteristics	
	Total Assets	
	Asset Allocations	
	Assumed Investment Rates of Return	
	Common Themes Among Member Associations	12
Chapter 2:	Kentucky Retirement Systems	13
-	Summary	13
	Membership	14
	Funding	17
	Statutes	17
	Governance	17
	Trustees	
	Administration	
	Fiduciary Responsibility	
	Liability and Insurance	
	Pension and Insurance Funds	
	Assets	
	Summary of Changes in Net Plan Assets	
	Asset Allocation Targets	
	Actual Allocations	
	Pension Fund	
	Insurance Fund	
	Investments	
	Objectives	
	Organization and Operation	
	Reporting	
	Internal and External Management	28

Chapter 3:	Kentucky Teachers' Retirement System	29
-	Summary	
	Membership	30
	Funding	31
	Statutes and Regulations	31
	Governance	32
	Trustees	32
	Fiduciary Responsibility	32
	Liability and Insurance	33
	Assets	
	Summary of Changes in Net Plan Assets	34
	Asset Allocation Targets	
	Actual Asset Allocations	
	Investments	
	Goals and Objectives	40
	Recommendation 3.1	40
	Organization and Operation	41
	Reporting	
	Internal and External Management	
Chapter 4:	Measuring Performance	45
	KRS and KTRS Investment Performance	45
	Summary	
	Benchmarks Provide Context	46
	Fund and Asset Class Benchmarks	47
	Kentucky Retirement Systems	47
	Recommendation 4.1	49
	Recommendation 4.2	49
	Kentucky Teachers' Retirement System	50
	Recommendation 4.3	
	Comparable States	51
	Recommendation 4.4	
	Actuarially Assumed Investment Rates of Return	
	Kentucky Retirement Systems	
	Kentucky Teachers' Retirement System	
	Comparable Systems	
	System Long-term Objective	54
	Kentucky Retirement Systems	
	Kentucky Teachers' Retirement System	
	Comparable Systems	
	Investment Performance of Comparable Systems	
	Fund Returns Relative to KRS.	
	Fund Returns Relative to KTRS	
	Asset Class Allocations and Returns Relative to KRS and KTRS	
	Allocations	
	Returns	

G	overnance and Investment Practices of Comparable Systems	
	Governing Bodies	.60
	Governance Requirements	.60
	Investing Factors	.60
F	actors Related to Investment Performance	
Works Cited		.65
Appendix A:	Balancing the Equation	.69
Appendix B:	Accounting Standards	.71
Appendix C:	Key Elements of Governmental Accounting Standards Board	
	Statements 43 and 45 Relative to KRS and KTRS Insurance Funds	.75
Appendix D:	Actuarial Assumptions and Benchmarks for Comparable Public	
11	Employee and Teacher Retirement Systems	.79
Appendix E:	Asset Allocations and 1-year and 5-year Investment Returns for KRS, KTRS, and Comparable Public Employee and Teacher Retirement Systems	
Appendix F:	Sources of Data for Comparable Public Employee and	
11	Teacher Retirement Systems	.83

List of Tables

2.1	Summary of Kentucky Retirement Systems' Changes in Net Plan Assets by	
	Revenue and Expense Categories for Pension and Insurance Funds,	
	Fiscal Years 2005-2007	.20
2.2	Acceptable and Prohibited Investments by Asset Class for KRS	.22
2.3	Target Asset Allocations KRS's Pension and Insurance Funds,	
	Fiscal Years 2002 and 2007	.23
3.1	Summary of KTRS's Changes in Net Plan Assets by Revenue and Expense	
	Categories, Fiscal Years 2005-2007, Pension and Insurance Funds	.34
3.2	KTRS Investment Category and Asset Allocation Limits	
3.3	Target Asset Allocations for KTRS Pension and Insurance Funds,	
	Fiscal Years 2007, 2008, and 2009	.36
3.4	KTRS Fixed-income Investment Guidelines	.38
3.5	KTRS Equity Investment Guidelines	.38
4.1	Summary of Actual Investment Performance Compared With Performance Measures	
	Approved by KRS and KTRS	.45
4.2	KRS's Reported Actual Rates of Return by Asset Class and Fund	
	Compared to Benchmarks	.48
4.3	KTRS Actual Rates of Return by Asset Class and Fund Compared to Benchmarks	.50
4.4	Benchmarks by Asset Class for 17 Comparable Systems, KRS, and KTRS,	
	Fiscal Year 2007	.52
4.5	KRS and KTRS Pension Fund Actuarially Assumed Rates of Return Compared to	
	Investment Returns	53
4.6	Current Actuarially Assumed Investment Rates of Return for Comparable	
	Public Employee or Teacher Retirement Systems	.54
4.7	Investment Rates of Return for State Public Employee Retirement Systems	
	With Active Membership Between 50,000 and 250,000	.56
4.8	Investment Rates of Return for State Teacher Retirement Systems With Active	
	Membership Between 50,000 and 100,000	.58
4.9	Asset Class Allocations for KRS, KTRS, and 22 Comparable Systems	.58
4.10	Selected Governance and Investment Characteristics for State Public Employee	
	Retirement Systems With Active Membership Between 50,000 and 250,000	.61
4.11	Selected Governance and Investment Characteristics for State Teacher Retirement	
	Systems With Active Membership Between 50,000 and 100,000	.62

List of Figures

1.A	Annual Rates of Return for Standard and Poor's 500 and 6-month
	U.S. Treasury Bills, Calendar Years 1988-2007
1.B	Total Financial Assets for State and Local Public Employee Retirement Systems,
	Calendar Years 1975-2006
1.C	Cash and Investment Holdings of U.S. Public Employee State and Local Retirement
	Systems, Fiscal Years 1987-200610
2.A	Total Kentucky Retirement Systems' Membership by Member Classification,
	Fiscal Years 2001-2007
2.B	Ratio of Active to Retired Members by System for the Kentucky Retirement
	Systems, Fiscal Years 2001-200716
2.C	Kentucky Retirement Systems' Pension and Insurance Funds Net Plan Assets and
	Pension Fund Percent of Total, Fiscal Years 2002-2007
2.D	KRS Pension Fund Actual and Target Asset Allocations, Fiscal Years 2002 and 200724
2.E	KRS's Insurance Fund Actual and Target Asset Allocations,
	Fiscal Years 2002 and 2007
2.F	KRS Investment Organization
3.A	Total KTRS Membership by Member Classification, Fiscal Years 2002-200730
3.B	Net Plan Assets, KTRS Pension and Insurance, Fiscal Years 2002-2007
3.C	KTRS Actual Asset Allocations for Pension and Insurance Funds,
	Fiscal Years 2000 and 2007
3.D	KTRS Investment Organization

Summary

At its October 2007 meeting, the Program Review and Investigations Committee voted to initiate a study of the investment practices and rates of return at the Kentucky Retirement Systems (KRS) and the Kentucky Teachers' Retirement System (KTRS). This report examines how KRS and KTRS investments are governed and managed. It also considers investment performance based on benchmarks established by KRS and KTRS and in comparison to other public employee and teacher retirement systems.

This report makes five recommendations.

Membership

KRS is responsible for investing funds and managing benefits for more than 316,000 active, retired, and inactive state and local government employees. It consists of three separate retirement systems: Kentucky Employees Retirement System, County Employees Retirement System, and the State Police Retirement System. Each system has separate pension and insurance funds.

KTRS has more than 119,000 members from employers from every public school district, five regional universities, the Kentucky Community and Technical College System, Department of Education agencies, and others. Per statute, eligible participants include certified employees and graduates of a 4-year college or university. Members are not eligible for federal Social Security benefits.

Governance

Nine-member boards of trustees for KRS and KTRS have ultimate responsibility for investments. Trustees are elected, appointed, or ex officio. The "prudent person rule," which states that trustees must use the same care, skill, prudence, and diligence that a prudent person with a similar capacity and in a similar environment would use, applies to both KTRS and KRS.

Each retirement system has an investment committee to provide regular investment oversight. KRS's investment committee consists of five board members. KTRS's committee is made up of two board members and the executive secretary. Investment expertise is not required for members of either committee.

KRS and KTRS employ professional staffs to manage investments and advise their boards. Investment consultants are hired to help each retirement system plan and allocate assets. Contracts with external fund managers allow each retirement system to diversify assets by selecting managers with specific investment expertise.

Assets

KRS manages \$17 billion in assets, and KTRS manages more than \$15 billion. Funding comes from employer contributions; member contributions; investment income, which consists of interest income and dividends and appreciation of assets; and other revenues. Of these sources, investment income has been largest. In fiscal year 2007, investment income totaled \$2.3 billion, or 72 percent of the value of additions to KRS, and \$2.0 billion, or 68 percent, to KTRS.

Other revenues have recently included Kentucky general fund appropriations. KTRS received \$12 million in FY 2007 and \$14 million in FY 2008. KRS received \$12 million in FY 2007. These appropriations were separate from regular employer contributions made by the Commonwealth of Kentucky to either KRS or KTRS.

Investments

KRS and KTRS expect to operate in perpetuity. As a result, both retirement systems hold long-run investment views. Short-term or periodic fluctuations in investment rates of return are mainly important for how they affect a retirement system's ability to fund retirement benefits in the long run. KRS pension and insurance funds are combined for investing purposes. KTRS also invests pension and insurance assets jointly.

KRS and KTRS each make investment decisions according to various factors, which include ensuring sufficient liquidity to pay benefit expenses throughout the fiscal year.

Investment goals establish the broad principles on which investment policies and procedures are established. Investment objectives are specific measures of those goals. KRS's investment goal is to "preserv[e] capital, while seeking means of enhancing revenues and protecting against undue losses in any particular investment area." Its objectives are to beat comparable unmanaged market indices in the short run and, in the long run, to exceed the actuarially assumed investment rate of return, which is 7.75 percent. KTRS identifies two "goals," which appear to be what are usually defined as objectives and two investment "concerns," which appear to be what are usually considered goals. KTRS lists two investment "objectives," but neither can be measured.

Recommendation 3.1

The Kentucky Teachers' Retirement System should report specific investment goals and measurable investment objectives.

Asset Allocations

KRS and KTRS invest in various financial instruments including domestic and international equities, fixed-income securities such as U.S. Treasury notes and bonds, and alternative investments such as real estate and private equity. The proportions of assets invested in these categories is a retirement system's asset allocation. Asset allocations are developed according to recommendations from investment consultants and staff, consideration of statutory and regulatory investment restrictions, and a system's investment philosophy.

Asset allocations play an important role in a retirement system's investment rate of return. All things equal, systems that invest proportionately more money in higher-risk investments achieve higher long-run returns than do other systems. Equities and alternative investments, in general, have more risk than fixed-income securities.

KRS and KTRS allocate assets differently. KRS's pension and insurance funds allocate assets separately but overall KRS allocates fewer assets to domestic equities and fixed-income securities than does KTRS and allocates more to international equities and alternative investments. Administrative regulations place numerous investment restrictions on KTRS; KRS's investments are not similarly restricted.

In recent fiscal years, KRS's and KTRS's asset allocations have changed. Both systems reduced their allocations to lower-risk investments such as fixed-income securities and increased their allocations to higher-risk, higher-return international equity and alternative investments. This corresponds to a similar allocation trend among other public employee retirement systems.

Investment Rates of Return

The table below shows KRS's and KTRS's investment rates of return over the most recent 1, 3, 5, and 10 years and net plan assets for FY 2007. At least part of the difference in rates of return is due to different asset allocations: higher rates of return generally correspond to higher risk asset allocations. Longer run returns are lower than 1- and 3-year returns, in large part, because they include declines in the domestic and international equity markets during the early 2000s.

	1	3	5	10	2007 Net
	Year	Year	Year	Year	Plan Assets
KRS Pension Fund	15.3%	11.4%	10.4%	8.1%	\$14.2 billion
KRS Insurance Fund	19.3%	13.7%	12.3%	8.4%	\$2.7 billion
KTRS Combined Funds	15.3%	9.3%	8.5%	7.1%	\$15.6 billion

KRS and KTRS Investment Rates of Return and Net Plan Assets

Sources: Commonwealth. Kentucky Retirement. *Comprehensive* FY 2007; Commonwealth. Teachers'. *Comprehensive* FY 2007.

Both KRS and KTRS approve multiple investment benchmarks that are used to evaluate asset class and fund-level investment performance. One benchmark is the actuarially assumed investment rate of return. This identifies the rate at which an actuary assumes investments will grow. KRS's most recent actuarial rate for its pension fund was 7.75 percent; KTRS's rate was 7.5 percent. Both rates have been higher in the past, so average rates over longer time periods will be higher than current rates. Returns for KRS and KTRS exceeded their actuarial rates over the past 1-, 3-, and 5-year periods but fell slightly short over the 10-year period.

Another benchmark is the retirement system's long-run objective. KTRS does not label this as an objective, but the system aims for an investment rate of return that exceeds inflation by 3.5 percentage points. For KRS, the long-run objective is the same as the current actuarially assumed rate of return. Actual returns for KRS and KTRS have exceeded these benchmarks.

A fund benchmark provides context for how well a retirement system's pension or insurance funds performed. The most straightforward fund benchmark is calculated by taking the benchmark indices used for each asset class within a fund and weighting them by each asset class's targeted or actual allocations as shares of the fund.

For KRS, actual reported returns for three of four pension asset classes and all four insurance asset classes for the most recent 1-year period were lower than their asset class benchmarks. The pension fund and the insurance fund, however, both exceeded their fund benchmarks. That KRS's 1-year actual returns exceeded the fund benchmarks while none or only one of the reported asset class benchmarks were met is counterintuitive and can be confusing. The problem is that investment returns for cash assets were not reported in KRS's *Comprehensive Annual Financial Report* for fiscal year 2007.

Recommendation 4.1

Subject to reporting standards, the Kentucky Retirement Systems should report investment returns and benchmarks for all classes of assets.

Another issue in comparing fund and asset class benchmarks is that KRS uses a customized fund benchmark. It would be helpful if supporting information were provided that clarified how these fund benchmarks are calculated, including how fund indices differ from asset class indices and how the weighting by asset class allocation is done.

Recommendation 4.2

The Kentucky Retirement Systems should report more detailed information on how its fund benchmarks are calculated.

Fund benchmarks were not reported in KTRS's *Comprehensive Annual Financial Report*, but KTRS staff provided these benchmarks to Program Review staff.

Recommendation 4.3

The Kentucky Teachers' Retirement System should report its fund benchmark, including a description of how that benchmark was calculated.

KTRS reports one combined benchmark for U.S. and international equities. The system has increased its investments in international equities, so it would be helpful if U.S. and international equity returns and benchmarks were reported separately.

Recommendation 4.4

The Kentucky Teachers' Retirement System should report separate U.S. and international equity asset class returns and benchmarks.

Other States

Knowing how well other retirement systems' investments performed can be useful when considering KRS's and KTRS's rates of return. Differences in investment policies, member demographics, funding, asset size, asset allocations, and other factors should be kept in mind when making comparisons, however.

Program Review staff identified and reported investment rates of return for 25 other public employee or teacher retirement systems with active membership numbers similar to KRS or KTRS. Average actual investment rates of return for those retirement systems exceeded KRS's pension fund over the 1-, 3-, 5-, and 10-year periods; KRS's insurance fund over the 10-year period; and KTRS's combined pension and insurance fund over the 1-, 3-, 5-, and 10-year periods. Consistent with academic research, these differences are attributable in some measure to different asset allocations. Higher long-run investment rates of return generally accompany higher-risk asset allocations. Judgments of decision makers may also play a role, including whether assets are invested internally or externally, how much risk is taken within a particular asset class, and which investment managers the system contracts with.

Program Review staff also identified differences in governance and investing among comparable public employee or teacher retirement systems. First, 10 out of 20 public employee retirement systems and 4 out of 9 teacher retirement systems examined in this report do not elect members to the board. Few systems are required by statute or regulation to operate an investment committee; however, most voluntarily established one. Most systems require board or investment committee members to have investment expertise, and two systems require investment training. Specific to investments, most comparable systems do not impose asset allocation limits, international investing restrictions, or in-state investing requirements.

Chapter 1

Overview and Background

Kentucky Retirement Systems (KRS) and Kentucky Teachers' Retirement System (KTRS) provide pension and health care benefits to qualified state and local government workers, teachers, and beneficiaries. KRS has \$17 billion and KTRS more than \$15 billion in net assets.	The Kentucky Retirement Systems (KRS) and the Kentucky Teachers' Retirement System (KTRS) provide pension and health care benefits to qualified state and local government employees, teachers, and their beneficiaries. Funding comes from employers' contributions, members' contributions, interest income and dividends, gains in asset values, and other sources. In total, KRS has \$17 billion and KTRS more than \$15 billion in net plan assets.
	KRS has more than 148,000 active members and more than 77,000 retired and beneficiary members in three separate systems: the Kentucky Employees Retirement System, the County Employees Retirement System, and the State Police Retirement System. KTRS has more than 75,000 active members and more than 39,000 retired and beneficiary members.
	The boards of trustees for KRS and KTRS have ultimate responsibility for investment of assets. Internal professional investment staff direct day-to-day investment operations and monitor external fund managers.
KRS and KTRS have diversified their portfolios in recent years.	In recent years, KRS and KTRS have diversified their investment portfolios and reallocated assets in order to improve returns and efficiently manage risk. Investment policies and asset allocation plans guided these changes. KTRS's allocations are, by statute, established in administrative regulations. KRS's investment allocations are approved by its board. KRS and KTRS maintain long-term investment views and have the powers and privileges of corporations.
As of June 30, 2007, the 10-year average return for KRS's pension fund was 8.1 percent and 8.4 percent for its insurance fund; KTRS's combined pension and insurance funds returned 7.1 percent.	Investment rates of return averaged 8.1 percent for KRS's pension fund and 8.4 percent for its insurance fund over the previous 10-year period ending June 30, 2007. For KTRS, the 10-year rate of return averaged 7.1 percent for its combined pension and insurance funds. Returns for both KRS and KTRS have been significantly higher in recent fiscal years.
	Long-run liabilities exceed assets for both KRS and KTRS. Actuarial funding levels for KRS range from 57 percent to 84 percent for its three pension funds and from 12 percent to

Actuarial funding levels remain below 85 percent, sometimes significantly so, of expected liabilities for KRS's and KTRS's various pension and insurance funds. Two percent of KTRS's estimated future insurance obligations have been funded. 50 percent for its three insurance funds. For KTRS, the actuarial funding level is 72 percent for its pension fund and 2 percent for its insurance fund.

Description of This Study

The study has two objectives: 1) describe the organization and operation of KRS and KTRS as related to investments; and 2) examine investment practices, including rates of return, for KRS and KTRS.

How This Study Was Conducted

Information for this report was gathered from interviews, a literature review, and analyses of data from various public employee and teacher retirement systems.

Program Review staff interviewed staff from the Kentucky Retirement Systems and the Kentucky Teachers' Retirement System. Representatives of the Kentucky Teachers' Retirement Association, Kentucky Education Association, Kentucky League of Cities, and the Kentucky Transportation Employees' Association were also interviewed. Program Review staff contacted the National Association of State Retirement Administrators and staff from Boston College's Center for Retirement Research.

Articles from peer-reviewed journals provided a detailed overview of governance and investment factors related to public employee retirement systems.

Staff also reviewed statutes and administrative regulations related to governance and investment practices of 25 comparable public employee and teacher retirement systems. These systems were identified on the basis of active membership. Investment rates of returns, asset allocations, investment expenses, and benchmarks were also analyzed.

Program Review staff analyzed KRS and KTRS annual reports, audits, and actuarial valuations; surveys by the National Association of State Retirement Administrators and the U.S. Census Bureau; and reports by the Wisconsin Legislative Council, U.S. Federal Reserve, Morningstar, Wilshire Consulting, and the PEW Charitable Trusts.

Organization of the Report

Chapter 1 introduces the study topic and presents seven major conclusions. Basic finance and investment principles related to retirement investments are described. A review of a federal law related to retirement systems follows. The next section describes public employee retirement system characteristics from a national perspective. The chapter concludes by identifying common themes noted by representatives of retirement system member associations.

Chapter 2 describes the Kentucky Retirement Systems. Chapter 3 describes the Kentucky Teachers' Retirement System. Each chapter describes membership, funding, and governance issues and concludes with a description of investment policies.

Chapter 4 considers investment rates of return of KRS and KTRS. Investment returns of comparable public employee and teacher retirement systems are also considered.

Appendix A describes the asset and liability equation for retirement systems. Appendix B describes important pension accounting standards. Appendix C describes important Governmental Accounting Standards Board statements as they relate to KRS's and KTRS's insurance funds. Appendix D details actuarial investment rate of return assumptions, long-term investment objectives, and policy objectives for comparable public employee retirement systems. Appendix E details asset allocations and investment rates of return for comparable public employee retirement systems. Appendix F lists the sources of information for the 25 comparable systems analyzed in this report. This report has seven major conclusions:

1. KRS and KTRS boards of trustees have ultimate fiduciary responsibility for investments, but both appoint investment committees and employ professional staff.

2. Risk-return decisions reflect retirement system circumstances and the judgment of decision makers.

3. Among retirement systems, higher-risk allocations, significant total assets, and lack of in-state investment requirements generally produce higher investment rates of return.

4. In the most recent 3 years, investment income comprised between 45 percent and 72 percent of the values of annual additions to KRS or KTRS assets.

5. In recent years, KRS and KTRS further diversified their portfolios and allocated more to international equities and other higher-risk investments, consistent with the national trend.

6. KRS and KTRS should report all asset class returns and benchmarks. KRS should explain its fund benchmarks, and KTRS should report and explain its fund benchmark.

7. KRS's and KTRS's investment rates of return met or exceeded most system-approved benchmarks. Only KRS's insurance fund had higher rates of return than the average of comparable systems.

Major Conclusions

This report has seven major conclusions.

- 1. Fiduciary responsibility for investments ultimately rests with the boards of trustees for KRS and KTRS. Both boards appoint investment committees to provide oversight and employ professional staff to monitor external fund managers and to perform day-to-day investment functions including compliance reviews of investment policies.
- 2. Typically, there is a trade-off between investment rates of return and risk. Because there is no single objective standard for what the acceptable trade-off should be, asset allocations, and thus rates of return, will vary according to the circumstances of each retirement system and the judgment of decision makers.
- 3. Among retirement systems, higher long-run investment rates of return are generally associated with higher-risk asset allocations. Significant total assets and lack of in-state investing requirements may also increase rates of return.
- 4. In the most recent 3 fiscal years, investment income, which includes interest, dividends, and appreciation of asset values, comprised between 45 percent and 72 percent of annual increases in value of KRS or KTRS assets.
- 5. In recent years, KRS and KTRS further diversified their investment portfolios and allocated more assets to international equities and other higher-risk investments, consistent with the national trend.
- 6. KRS and KTRS should report all asset class returns and benchmarks. KRS should explain its fund benchmarks, and KTRS should report and explain its fund benchmark.
- 7. Actual investment rates of return for different investment periods met or exceeded most benchmarks approved by KRS or KTRS. Only KRS's insurance fund had higher rates of return than the average of comparable public employee or teacher retirement systems.

Basic Finance and Investment Principles

Trade-off Between Risk and Return

Investments are typically categorized by their level of risk and corresponding rate of return. Generally, higher returns accompany higher risk, and lower returns go with lower risk. This pattern reflects investor demands for returns commensurate with risk.

Finance professionals describe an investment's level of risk by its volatility. Greater volatility indicates greater levels of risk, at least in the short run. Considered over longer periods of time, volatility is typically lower. Similarly, long-run investment returns are typically less volatile than short-run returns.

Historically, equities have been higher-risk investments. Annual returns, as measured by the Standard and Poor's 500, have fluctuated up to 30 percent or more by year. By comparison, U.S. Treasury bill returns, a relatively risk-free, short-term security, tended to vary little. Figure 1.A shows total annual returns for equities compared with 6-month U.S. Treasury bills. The historic rate of return for equities is about 10 percent, unadjusted for inflation, compared with 3 percent for U.S. Treasury bills (Munnell, Sass, and Soto).

Figure 1.A Annual Rates of Return for Standard and Poor's 500 and 6-month U.S. Treasury Bills, Calendar Years 1988-2007



Sources: U.S. Board of Governors of the Federal Reserve. "Selected"; Standard and Poor's; Mutual of America.

Higher returns generally accompany higher risk. For example, in the long run, higher-risk domestic securities, although prone to significant volatility, have outperformed lower-risk, stable U.S. Treasury bills.

Investment Is Long Term

Most public employee retirement systems expect to operate in perpetuity, so they hold long-term investment views. Fluctuations in annual or short-term rates of returns are mainly important for how they affect a system's ability to fund retirement benefits in the long run.

Diversification Reduces Risk

Retirement systems typically invest in a diverse group of assets, among a diverse group of investment managers. The purpose is to minimize risk associated with investing in a limited number of assets or asset classes. Through diversification, retirement systems reduce the risk that a particular asset's poor returns will inordinately weigh down overall rates of return.

Liquidity

Liquidity indicates how easily an asset can be sold without significant loss. Because public employee and teacher retirement systems incur regular benefit expenses, liquidity is an important investment consideration. Depending on their membership demographics, some systems will have to devote more assets to current beneficiaries than others and must adjust their investments accordingly. Having sufficient liquidity can reduce or eliminate the need to sell assets at a loss in order to pay retirement benefits or other expenses.

Employee Retirement Income Security Act

The 1974 Employee Retirement Income Security Act (ERISA) set minimum standards for most voluntarily established pension and medical plans in private industry. It does not cover governmentsponsored pension plans, but KRS and KTRS have established standards comparable to those required by ERISA.

ERISA requires plan sponsors to

- provide participants with plan information including features and funding;
- set minimum standards for participation, vesting, benefit accrual, and funding;
- provide fiduciary responsibilities for those who manage and control plan assets;

Public employee retirement systems maintain long-term investment perspectives.

Diversification can reduce risk and lead to more stable investment rates of return.

Liquidity indicates how easily an asset can be sold without significant loss.

The 1974 Employee Retirement Income Security Act established retirement plan standards for private industry. KRS and KTRS use similar standards.

- require the establishment of a grievance and appeals process for participants to get benefits from their plans; and
- give participants the right to sue for benefits and breaches of fiduciary duty.

National Perspective

Governments of all sizes and from every state operate retirement systems for the purpose of providing benefits to employees when they retire. Many retirement systems originated in the mid-1900s.

Most systems are defined benefit plans, which are designed to guarantee retirees a certain proportion of their income in retirement. In exchange for regular contributions, employees become eligible to receive pension, insurance, or disability benefits based on their incomes and lengths of service. State and local employers usually make matching, and sometimes higher, contributions. Contributions are then managed and invested by a retirement system or other governing body.

Governance

In most systems, the board of trustees has ultimate fiduciary responsibility for a retirement system's financial resources. Of greatest importance is how a board decides to invest its assets. Many systems follow language similar to the "prudent person" standard used under ERISA, which states that investment decisions should be made "solely in the interest of the participants and beneficiaries" and "with the same care, skill, prudence, and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters" would use (29 U.S.C. 1104(a)).

State constitutions, statutes, and regulations occasionally restrict the types of investments that boards can make (Mitchell et al). For example, until 2007, South Carolina's constitution prevented public employee retirement systems from investing in international equities ("South Carolina"). Indiana public employee retirement systems, in another example, could not invest in equities until 1996 (Useem and Hess). Overall, investment restrictions have eased in recent years, and public employee retirement systems now hold more diverse assets.

Retirement boards typically approve contracts with investment consultants, legal counsel, and actuaries (Steffen). Actuarial

Most retirement systems operate defined benefit pension plans whose assets are managed and invested by the system or other governing body.

Fiduciary responsibility for a retirement system's financial resources ultimately resides with the board of trustees. Most retirement systems employ a "prudent person" standard that requires trustees to use the same care, skill, prudence, and diligence as any other person acting in a similar capacity.

Constitutional, statutory, and regulatory provisions can restrict the types of investments a public employee retirement system can make. assumptions may be approved by boards. Board members generally hire an executive director and establish rules for the operation and administration of the retirement system (Steffen). Budgets for the majority of state public employee retirement systems are approved by the state legislature, but about one-third are approved solely by the board (Hsin).

Trustee authority over investments varies. Most boards oversee investment policy, but some have limited roles (Useem and Hess).

Most boards consist of 8 trustees, but some have as many as 18 or as few as 4 (Hsin). Board representation can include members elected by plan participants or beneficiaries, appointed officials, or elected officials who automatically serve as ex officio members. State statute generally defines representation.

General Characteristics

According to a U.S. Census Bureau survey, there are at least 134 state-level public employee or teacher retirement systems in 39 states (U.S. Census. *State*).

State-level public employee or teacher systems vary in membership and asset size. Some systems provide benefits to fewer than 5,000 members and manage less than \$400 million. Some retirement systems provide benefits to as many as 500,000 members and manage \$200-billion portfolios. As of 2006, of public employee or teacher retirement systems, the median size was 79,100 active members; 38,300 retired members; and \$12.7 billion in assets (U.S. Census. *State*).

Most retirement systems have sufficient financial resources to pay current benefits. However, over the long run, actuarially determined liabilities exceed assets for most systems. This means current contributions and assets are not sufficient to pay future obligations. According to a recent report, the average state public employee pension system was approximately 85 percent funded. Medical insurance benefits were mostly unfunded (Pew Center).

Total Assets

Total assets indicate how much is available to pay liabilities. It is a key component in measuring a retirement system's funding adequacy, setting employer and member contribution rates, and making assumptions about future investment income.

Most boards of trustees have eight members, of which some are elected, appointed, or ex officio.

As of 2006, the median state public employee retirement system had 79,100 active members and 38,300 retired members and \$12.7 billion in assets.

On average, most retirement systems have sufficient financial resources to pay current benefits but only about 85 percent of longrun liabilities.

Total assets indicate the value of resources available to pay liabilities and are a factor in making assumptions about future investment income. Total assets for state and local public employee retirement systems increased, on average, 12 percent per year from 1975 to 2006.

Asset values can affect asset allocation decisions which, in turn, can affect investment rates of return. Figure 1.B shows total assets belonging to state and local public employee retirement systems rising from \$104 billion in 1975 to \$3.1 trillion in 2006. Although total assets increased, on average, 12 percent per year, total assets fell in 2000, 2001, and 2002. The dot-com bubble, terrorist attacks, and subsequent recession contributed to the decline.

Asset values can affect asset allocation decisions and, in turn, investment returns. For example, a retirement system whose asset value exactly matched expected current-year expenses would invest in short-term securities because all assets would be needed in the near term, assuming no net plan additions during the year.

Figure 1.B Total Financial Assets for State and Local Public Employee Retirement Systems, Calendar Years 1975-2006



Source: U.S. Board of Governors. "Flow of Funds Accounts."

Asset values increase whenever interest income, dividends, employer contributions, member contributions, or the market value of investments increase. Asset values decrease whenever benefits or investment expenses increase, or investments lose market value. All else equal, asset values increase when interest income, dividends, employer contributions, or member contributions increase. They also rise when the market value of investments increases. Asset values fall whenever benefit or investment expenses rise or investments lose market value. Appendix A has more detail.

Asset Allocations

Asset allocations are the mix of investments that each system carries.

The way in which systems allocate assets can account for as much as 92 percent of the differences in systems' total return from their investment portfolios. Retirement systems invest in stocks, bonds, real estate, and other financial instruments. The mix of particular investments that each system carries is its asset allocation. Diversified systems hold a broad range of investments, which should produce more stable rates of return.

Asset allocation is also important because it is the primary determinant of investment return. Research has shown that the way in which systems allocate assets can account for as much as 92 percent of the differences in systems' total return from their investment portfolios (Brinson; Ambachtsheer; Schneider). For example, a retirement system that invests primarily in low-risk government securities will likely realize lower rates of return over the long run compared to a system that invests in higher-risk, higher-return domestic equities.

In recent years, public employee and teacher retirement systems significantly altered their asset allocations. Figure 1.C shows asset allocations for state and local public employee retirement systems from 1987-2006.





Notes: "Cash and deposits" include various short-term instruments invested for as long as 1 year. "Governmental securities" include U.S. Treasury, federal agency, and state and local government securities. "Nongovernmental securities" include primarily domestic stocks and bonds but also mortgages and foreign and international securities. "Other investments" include, among others, real property.

Sources: 1987-2002 data from U.S. Census "Employee"; 2003-2006 data from U.S. Census. *State*.

Public employee retirement systems have been shifting investments from fixed income securities and toward domestic and international equities.

Assumed rates of return are used to estimate future investment income.

On average, assumed rates of return have not increased in recent years despite retirement systems reallocating assets into higher-risk securities. In 1987, public employee and teacher retirement systems held less than 60 percent of assets in nongovernmental securities such as domestic and international equities and more than 25 percent in governmental securities such as U.S. Treasury obligations. By 2006, allocations were about 80 percent nongovernmental securities and less than 10 percent governmental securities.

Within this broad investment change, state and local public employee and teacher retirement systems significantly expanded investments in foreign and international securities. International stock allocations rose from 0.3 percent in 1990 to 6.7 percent in 1996 (Eaton). More recently, foreign and international investments increased from 11.8 percent in 2002 to 15.0 percent in 2006, a 27 percent rise (U.S. Census. *State*).

Assumed Investment Rates of Return

Assumed investment rates of return are figures developed by actuaries and approved by boards to help retirement systems gauge, among other things, future investment income.

Frequent revisions to assumed rates of return occur. Retirement systems modify assumptions based on changes in market performance. Some boards may evaluate and change assumed rates as a regular course of action.

Despite retirement systems reallocating some assets from lowerrisk governmental securities to higher-risk international securities, assumed rates of return have typically not increased. The average assumed rate of return for state and local public employee retirement systems was 8.03 percent in 2000 but only 7.95 percent in 2006 (State of Wisconsin).¹ The median rate of return, or the amount at which one-half of systems had higher and one-half had lower rates, remained constant at 8 percent. Potential reasons include a lag between asset allocation adjustments and retirement system changes to assumed rates of return, retirement systems' attachment to long-held expectations, retirement systems' desire to maintain a conservative estimate, or other external influences (Hustead).²

¹ In 2006, the highest assumed rate was 9 percent and the lowest was 6.5 percent.

² Expected rates of return increased from 3 percent to 4 percent in the 1950s and 1960s—a time when public employee retirement systems were largely invested in governmental securities—to around 8 percent in the early 1990s, when retirement systems redirected some of their assets into nongovernmental securities (Hustead).

Common Themes Among Member Associations

Program Review staff interviews with representatives of public employee and teacher associations revealed three common themes: concerns about underfunded pension and medical benefits, dissatisfaction with limited board representation, and satisfaction with the way KRS and KTRS investments are managed.

Concerns that KRS and KTRS may have inadequate resources to meet future obligations were common. Association representatives consistently mentioned funding as an important issue. Insufficient board of trustee representation was a concern for some. Overall, interviewees expressed satisfaction with the way KRS and KTRS operate, including their investment approaches and rates of return.

Chapter 2

Kentucky Retirement Systems

This chapter describes the Kentucky Retirement Systems. It begins with a summary and continues by describing membership, funding, and governance. The chapter concludes with a description of investment policies and practices.

Summary

KRS invests funds and manages benefits for more than 316,000 active, retired, and inactive state and local government employees. KRS consists of three separate systems: Kentucky Employees Retirement System (KERS), County Employees Retirement System (CERS), and the State Police Retirement System (SPRS). Each system has separate pension and insurance funds.

This chapter has four main points.

1. KRS is governed by a ninemember board of trustees. A five-member investment committee oversees investments.

2. Assets grew \$4.9 billion, or 40 percent, from FY 2002 to FY 2007.

3. KRS is changing its asset allocations to include a greater percentage of higher-risk, higherreturn securities, such as international equities.

4. Investment income accounted for 64 percent to 72 percent of annual net plan additions from FY 2005 to FY 2007. KRS is a state agency responsible for investing funds and managing benefits for more than 316,000 active, retired, and inactive state and local government employees. It is made up of three separate retirement systems: the Kentucky Employees Retirement System (KERS), the County Employees Retirement System (CERS), and the State Police Retirement System (SPRS). Each system has separate pension and insurance funds.

KERS was created in 1956 for the purpose of supplementing Social Security benefits for retired state government workers. CERS, a system primarily for county and city employees, and SPRS, a system exclusively for state police, were created in 1958 (Commonwealth. Kentucky Retirement. *Comprehensive* FY 2007).

This chapter has four main points.

First, KRS is governed by a nine-member board of trustees. Five members are elected, three are appointed, and one is ex officio. Members are not required to have investment expertise. A fivemember investment committee of board members oversees all investment activities.

Second, net assets grew \$4.9 billion, or 40 percent, from FY 2002 to FY 2007. KRS had \$17 billion in net assets at the end of FY 2007.

Third, KRS is changing its asset allocations by increasing its investments in international securities and private equity while decreasing fixed-income and short-term investments. Board approval is required to change asset allocations.

Fourth, investment income, which includes interest income, dividends, and appreciation of asset values, accounted for

64 percent to 72 percent of annual net plan additions from FY 2005 to FY 2007. Employer and member contributions were the next two largest components. Other revenue, including a \$12 million general fund appropriation in FY 2007, represented the balance.

Membership

Each system in KRS consists of active, retired, and inactive members. Active members are currently employed. Retired members include former employees receiving benefits and beneficiaries of deceased members. Inactive members no longer work for a participating agency.

Members of KERS include employees of state agencies and all regional universities. Total active membership in FY 2007 was 52,000. Retired and inactive members totaled 70,000.

CERS members include area development districts, boards of education, and city and county government employees. Its membership is the largest of the three systems with 95,000 active members and 97,000 retired or inactive members.

SPRS members are State Police employees. It is the smallest of the three systems, with 957 active members and 1,400 retired or inactive members.

For the KERS and CERS systems, there are two types of members: hazardous and nonhazardous. Hazardous duty members typically perform more dangerous or risky jobs. The SPRS system is hazardous duty only.

Among others, members of KERS include employees of state agencies and regional universities. CERS members include area development district and city and county government employees. SPRS members are limited to State Police employees. Program Review and Investigations

From FY 2001 to FY 2007, KRS active membership rose 6 percent, from 140,000 to 148,000. Retired membership rose 43 percent, from 54,600 to 77,900. The ratio of active to retired members has steadily declined.

Figure 2.A shows total KRS membership increasing from 241,800 in FY 2001 to 316,100 in FY 2007. Active membership increased 6 percent from 140,000 to 148,000; retired membership increased 43 percent from 54,600 to 77,900; and inactive membership increased 93 percent from 46,700 to 90,100.

Figure 2.A Total Kentucky Retirement Systems' Membership by Member Classification, Fiscal Years 2001-2007



Source: Staff's compilation of information from Commonwealth. Kentucky Retirement. *Comprehensive* FY 2001-FY 2007.

Figure 2.B shows the ratio of active to retired members for all three KRS systems and hazardous/nonhazardous classifications. Active-to-retired ratios typically express the concentration of retirement costs among current employees. Higher ratios indicate the financial burden is more dispersed, which means lower perperson burdens. Lower ratios indicate costs are spread among fewer active members.

The ratio of active to retired KRS members has declined from 2.6 to 1 in FY 2001 to 1.9 to 1 in FY 2007. Each system experienced similar decreases. The most dramatic was for the KERS hazardous duty system, whose ratio fell from 4 to 1 active members per retiree or beneficiary to 2 to 1.

Figure 2.B Ratio of Active to Retired Members by System for the Kentucky Retirement Systems, Fiscal Years 2001-2007



Source: Staff's analysis of Commonwealth. Kentucky Retirement. *Comprehensive* FY 2001-FY 2007.

Funding

KRS's primary revenue sources are employer contributions, member contributions, interest income and dividends, and net asset appreciation.

Other revenue in recent years has included \$48 million from Anthem Insurance demutualization, \$21.5 million from Medicare, and \$12 million from Kentucky's general fund.

Member contribution rates are set by statute. Employer contribution rates for KERS and SPRS are subject to legislative approval, but CERS employer rates are not.

The KRS Board of Trustees includes five elected members, three appointed members, and one ex officio member.

KRS receives revenue from four primary sources: employer contributions, member contributions, interest income and dividends, and net asset appreciation. Other revenue, such as state appropriations and other one-time sources, is significantly smaller.

KRS received \$48 million from the demutualization of Anthem Insurance, \$21.5 million from the Medicare Retiree Drug Subsidy Program, and \$12 million from the state general fund in recent years (Commonwealth. Kentucky Retirement. "Kentucky"). The latter was appropriated by House Bill 380, the 2006-2008 biennial budget, and used to shore up the Kentucky Retirement Systems' insurance fund. These were one-time revenues.

Member contribution rates are set by statute. For KERS and SPRS, employer contribution rates are determined by the board of trustees based on annual valuations, subject to legislative approval. Employer contribution rates for CERS are determined by the board of trustees without further legislative review.

Statutes

Statutory authority for the Kentucky Retirement Systems comes under KRS Chapter 61. KRS 61.645 covers the board of trustees' composition, selection, powers, terms, fiduciary responsibilities, liability, and required reports. KRS 61.650 covers the investment committee and fiduciary duties of officers and employees.

Governance

Trustees

KRS is governed by a nine-member board of trustees. Two trustees are elected by KERS members, two are elected by CERS members, one is elected by SPRS members, three are appointed by the governor, and the secretary of the Personnel Cabinet serves as an ex officio member. At least one gubernatorial appointee must be knowledgeable about the impact of pension requirements on local governments. Each trustee's term is 4 years. Trustees are paid through per diems.

Administration

The board appoints or contracts for the services of an executive director. As the chief administrative officer of the board, the executive director oversees day-to-day operations of the system. The board retains exclusive power and control over employees and meets at least quarterly.

An investment committee of five board members, each appointed by the board chair, oversees assets. Acting on behalf of the board, the committee has the power to acquire, safeguard, monitor, and manage system assets and securities. This includes reviewing reports and approving the selection and termination of service providers (Commonwealth. Kentucky Retirement. Board. Sec. 2.2, d. 1.). Meetings are held at least quarterly. Members are not required to have investment expertise.

Fiduciary Responsibility

Statute requires that trustees, as fiduciaries, discharge their duties in good faith, on an informed basis, and in a manner honestly believed to be in KRS's best interests. Officers, employees, and other fiduciaries must also work for the sole interest of KRS's members and beneficiaries.

Liability and Insurance

Suits against KRS trustees for monetary damages may only be brought for acting or failing to act on a matter where the trustee's behavior constitutes willful misconduct or wanton or reckless disregard for human rights, safety, or property. This does not alleviate the need for insurance.

The executive director is covered by a surety bond of \$500,000. A Public Entity Fiduciary Liability Policy of \$5 million insures the pension and insurance trusts and any past, present, or future trustee or employee of the trusts. Premiums are paid by the KRS administrative budget.

Pension and Insurance Funds

The Kentucky Retirement Systems' board administers separate funds on behalf of its member systems—a pension fund and an insurance fund—but combines the funds for purposes of investing. Each has a separate written investment policy incorporated in the bylaws. The pension fund invests the pension assets of KERS,

A five-member investment committee appointed by the board chair oversees investments. Committee members are not required to have investment expertise.

Statute mandates that trustees, officers, employers, and others complete their duties in good faith and in the best interest of KRS.

KRS administers separate pension and insurance funds.
CERS, and SPRS (Commonwealth. Kentucky Retirement. "Statement of Investment Policy—Pension"). Statute establishes the Kentucky Retirement Systems' Insurance Fund, which is to provide fringe benefits to recipients of KERS, CERS, and SPRS (Commonwealth. Kentucky Retirement. "Statement of Investment Policy—Insurance"). It is administered in the same manner as the pension funds (KRS 61.701).

Assets

Between FY 2002 and FY 2007, net plan assets grew from about \$12 billion to almost \$17 billion, a 40 percent increase. Of KRS's net plan assets, the pension fund holds 84 percent and the insurance fund holds 16 percent. Figure 2.C shows net plan assets for KRS's pension and insurance funds. Net plan assets reflect contributions received, interest income and dividends earned, and asset appreciation less currentyear benefit payments and other expenses. Between FY 2002 and FY 2007, net plan asset values grew from approximately \$12 billion to almost \$17 billion, a 40 percent increase, or 6.7 percent per year. Pension fund assets relative to total net assets declined from 90 percent in FY 2002 to 84 percent in FY 2007. Insurance fund assets increased by a corresponding amount.





Source: Staff's analysis of Commonwealth. Kentucky Retirement. *Comprehensive* FY 2002-FY 2007.

Summary of Changes in Net Plan Assets

Table 2.1 identifies the major components of annual net plan asset changes. They are divided into revenue and expense categories. Revenues include employer and member contributions, net investment appreciation, and interest income and dividends. Reported amounts combine KRS's pension and insurance funds for simplicity.

Table 2.1 Summary of Kentucky Retirement Systems' Changes in Net Plan Assets by Revenue and Expense Categories for Pension and Insurance Funds, Fiscal Years 2005-2007 (in thousands of dollars)

	FY 2005		FY 2006		FY 200	7
		% of Sub-		% of Sub-		% of Sub-
	Total	total	Total	total	Total	total
Revenues						
Member contributions	\$310,806	16%	\$288,230	13%	\$326,876	10%
Employer contributions	\$390,189	20%	\$469,935	22%	\$601,162	18%
Net investment appreciation	\$845,957	44%	\$991,827	46%	\$1,874,664	57%
Interest and dividends	\$381,329	20%	\$413,592	19%	\$478,171	15%
Subtotal	\$1,928,281	100%	\$2,163,584	100%	\$3,280,873	100%
Expenses						
Benefits	\$1,127,413	97%	\$1,275,396	97%	\$1,428,203	97%
Administrative	\$18,663	2%	\$23,931	2%	\$27,824	2%
Investment	\$17,947	2%	\$21,279	2%	\$21,883	1%
Subtotal	\$1,164,023	100%	\$1,320,606	100%	\$1,477,910	100%
Additions less deductions	\$764,258		\$842,978		\$1,802,963	
Other changes	\$(11,935)		\$(24,038)		\$3,702	
Net plan asset change	\$752,323		\$818,940		\$1,806,665	

Note: "Other changes" include capital expenses. Subtotals may not add to 100% due to rounding. Source: Compiled by staff based on Commonwealth. Kentucky Retirement. *Comprehensive* FY 2005-FY 2007.

Net plan assets increased primarily due to investment income, which is appreciation of investments plus interest income and dividends. Net plan assets increased in recent years primarily due to appreciation of investments and positive interest income and dividends. Since FY 2005, these factors accounted for 64 percent to 72 percent of KRS's net plan asset additions per year. Employer and employee contributions represented about one-third of additions. Benefit payments account for almost all expenses.

Investment expenses include common stock commissions, broker rebates, and contractual service costs.

Asset allocation targets establish how much should be invested in particular asset classes.

State statutes and administrative regulations do not specifically prohibit specific investments, but KRS imposes some restrictions of its own.

In-state investments are not required.

Since FY 2005, benefit payments accounted for approximately 97 percent of KRS's expenses. In FY 2007, pension benefits totaled \$1.19 billion of \$1.23 billion in total expenses. For the insurance fund, health care costs accounted for \$241 million of \$249 million in total expenses. Administrative and investment expenses comprised the balance.

Investment expenses include common stock commissions, broker rebates, and contractual service costs such as investment management.¹ Investment expenses are allocated in proportion to the number of active members participating in each plan and the carrying value of plan investments, respectively.

Asset Allocation Targets

Asset allocation targets establish how much should be invested in particular asset classes according to projected total assets, liabilities, and system goals and objectives. An investment consultant guides the allocation process.

Neither statute nor administrative regulation prohibits specific investments. However, KRS imposes its own restrictions based on its annually updated 5-year investment plan (Commonwealth. Kentucky Retirement. KRS). Table 2.2 identifies acceptable and restricted investments by asset class.

Statute requires the Kentucky Retirement Systems to give priority to investments expected to "improve the industrial development and enhance the economic welfare of the Commonwealth" whenever consistent with fiduciary responsibilities (KRS 61.650(3)). The board, however, is not required to make instate investments.

After considering investment policies, an investment consultant models liabilities, projects assets, determines total expected funding obligations, and considers multiple risk and reward scenarios. The consultant then recommends an asset allocation policy to KRS. The most recent asset allocation and liability study was in 2006 and considered a 5-year implementation timeline.

¹ Professional consultant fees to KRS's actuary, auditor, legal counsel, and similar other entities are not included in investment expenses. They are identified separately in the *Comprehensive Annual Financial Report* and totaled \$1.8 million in FY 2007.

Asset Class	Acceptable Investments	Prohibited Investments or Other Restrictions
Alternative/	Venture capital, private equity, and	Subject to specific investment
Real Estate	private placements	committee approval
Cash	Publicly traded investment grade	Must have BBB rating or above
Equivalent	corporate bonds, government and agency	by at least one bond rating service;
Securities	bonds, mortgages, collective short-term	maturity cannot exceed 2 years;
	investment funds, money market funds or	repurchase agreement maturity
	instruments such as certificates of	equal to period remaining until
	deposit, and repurchase agreements	repurchase of underlying security
Derivatives	Collateralized mortgage obligations and	Cannot leverage system beyond
	other asset-backed securities; exchange-	100 percent of position; large or
	traded funds, which convert cash into	unanticipated changes in duration
	short-term equity investments	or cash flow prohibited
Equities	Domestic and international common	
	stock, securities convertible into common	
	stock, and preferred stock of publicly	
	traded corporations	
Fixed Income	Publicly traded corporate bonds that	
	balance quality and maturity given	
	current market and economic conditions;	
	debt issued or guaranteed by agency or	
	instrumentality of U.S. government	
Mortgages	Direct real estate mortgages or mortgage	
	pools guaranteed by U.S. government	
	agency or Commonwealth of Kentucky	

 Table 2.2

 Acceptable and Prohibited Investments by Asset Class for KRS

Source: Commonwealth. Kentucky Retirement. Comprehensive FY 2007.

Asset allocation targets have decreased for domestic equity and fixed-income securities and increased for international equity and alternative investments. Table 2.3 shows pension and insurance fund asset allocation targets for FY 2007. It also shows targets for FY 2002, the period preceding the latest allocation study in 2006.

	Per	nsion	Insurance		
Asset Class	2002	2007	2002	2007	
Domestic equity	40%	30%	60%	40%	
International equity	15	20	20	30	
Fixed-income	27	25	0	0	
Alternatives	5	12	5	15	
TIPs	10	10	10	12	
Cash	3	3	5	3	
Total	100%	100%	100%	100%	

Table 2.3Target Asset Allocations for KRS's Pension andInsurance Funds, Fiscal Years 2002 and 2007

Note: TIPs is Treasury Inflation-protected Securities.

Source: Staff's compilation of information from Commonwealth. Kentucky Retirement. *Comprehensive* FY 2002 and FY 2007.

For the pension fund, the target for domestic equities decreased from 40 percent to 30 percent. The target for international equities increased from 15 percent to 20 percent. Alternative investment targets increased from 5 percent to 12 percent.

KRS's insurance fund, in general, maintains a higher-risk investment portfolio relative to KRS's pension fund.

Actual and target asset allocations do not usually match due to market fluctuations and regular asset management. Compared to the pension fund, the insurance fund had a more aggressive investment strategy but showed similar asset allocation changes. Domestic equity targets decreased from 60 percent to 40 percent of total allocations, and international equity rose from 20 percent to 30 percent. Target allocations for alternative investments increased from 5 percent to 15 percent. According to KRS staff, the insurance fund is appropriately diversified without holding any fixed-income assets (Burnside).

Actual investment practice usually does not match allocation targets. Frequent market fluctuations, along with regular, active asset management, create differences. Consequently, KRS investment staff monitor investment practices to ensure that allocations remain close to asset targets.

Actual Allocations

Figures 2.D and 2.E show actual asset allocations for KRS's pension and insurance funds for FY 2002 and FY 2007. KRS established a 5-year timeline to implement these asset targets. According to KRS staff, the system had 0.91 percent exposure to sub-prime securities as of December 31, 2007 (Burnside).

Pension Fund. For the pension fund, domestic and international equity holdings increased between FY 2002 and FY 2007. Although the FY 2007 domestic equity target is lower than it was in FY 2002, actual allocations are higher, due, in part, to the time it takes to make large allocation changes. A 10-point change in domestic equity pension fund target, for example, equals approximately \$500 million. According to KRS staff, changing asset allocations is similar to changing the direction of a super-tanker—it takes a long time.

Figure 2.D KRS Pension Fund Actual and Target Asset Allocations, Fiscal Years 2002 and 2007



Note: TIPs is Treasury Inflation-protected Securities.

Source: Staff's compilation of information from Commonwealth. Kentucky Retirement. *Comprehensive* FY 2002 and FY 2007.

Actual allocations for fixed-income decreased from 31 percent to 23 percent. Over the long run, fixed-income assets generally have lower returns compared to equities, in large part, because they are less volatile.

Significant deviations from allocation targets require KRS staff to rebalance assets.

KRS's asset allocations have generally followed national trends.

Overall, actual asset allocations for the Kentucky Retirement Systems' pension fund generally followed national trends. KRS moved some assets from fixed-income and cash to domestic and international equities, as many other state and local public employee retirement systems did.

Insurance Fund. For the insurance fund, increased allocations to domestic and international equities accounted for most change between fiscal years 2002 and 2007. These increases were offset by decreases to alternative and cash investments.

Figure 2.E KRS's Insurance Fund Actual and Target Asset Allocations, Fiscal Years 2002 and 2007



Note: TIPS is Treasury Inflation-Protected Securities. Source: Staff's compilation of information from Commonwealth. Kentucky Retirement. *Comprehensive* FY 2002 and FY 2007.

Investments

Goals

KRS's primary goal is to preserve capital while trying to enhance revenues.

KRS invests assets to provide retirement benefits to its members. According to its 2007 *Comprehensive Annual Financial Report*, the primary goal is to "preserv[e] capital, while seeking means of enhancing revenues and protecting against undue losses in any particular investment area." KRS staff also noted that they try to follow three related principles: diversification, globalization, and opportunity.

Objectives

Approved by the KRS Board of Trustees, investment objectives establish how retirement system investments will be evaluated in the short term and long term in relation to investment goals. Investment consultants advise the board on investment strategies, including asset allocations.

Since FY 2002, the board has held the same short-term objective: year-over-year returns for managed funds should exceed comparable unmanaged market indices. The long-term objective has changed. In FY 2002, long-term returns were expected to exceed the Consumer Price Index by at least 4 percentage points over two market cycles, about 6 to 10 years. In FY 2007, long-term returns on total assets were expected to exceed the actuarially assumed investment rate of return of 7.75 percent and blended performance benchmarks over two market cycles. In 2006, the board changed investment consultants.

Organization and Operation

Figure 2.F depicts the organizational structure for KRS's investments. All decisions ultimately reside with the board of trustees. A five-member Investment Committee oversees investment activities. An appointed executive director manages the retirement system, including investments, and reports to the board.

Ten investment staff monitor and evaluate internal and external investments. Three directors—equities, alternative investments, and fixed assets—report to the chief investment officer. A compliance officer monitors all portfolios to ensure they operate within acceptable allocation targets and follow investment policies and procedures. The compliance officer reports to the chief investment officer and the Investment Committee.

Investment staff from KRS perform multiple functions including managing investments, monitoring and evaluating external fund managers, rebalancing assets, examining and evaluating investment opportunities, monitoring compliance with investment policy, and developing reports. Staff also negotiate and execute contracts with service providers.

The short-term investment objective is to beat comparable unmanaged market indices. The long-term investment objective is to exceed actuarially assumed investment rate of return, which is currently 7.75 percent, and blended performance benchmarks.

The board of trustees has ultimate responsibility and authority to oversee investments.

Ten investment staff monitor and evaluate internal and external investments.

Monthly compliance tests ensure that asset allocations and risk exposures are acceptable. KRS investment staff conduct quarterly performance reviews of external fund managers. Monthly compliance tests are performed by the compliance officer to ensure that asset allocations and risk exposures are acceptable (Commonwealth. Kentucky Retirement. "Statement of Investment Policy—Pension" and "Statement of Investment Policy— Insurance"). Staff also conduct quarterly performance reviews that examine how well external fund managers performed relative to their benchmarks.

Staff also rebalance assets by continually monitoring and adjusting asset holdings to approximate asset allocation targets. Target differences of plus or minus 1 percentage point for a particular asset category may be adjusted. Whenever an asset class is within plus or minus 1 percentage point of the upper or lower target limit, staff must reallocate assets to ensure compliance with allocation targets.

Figure 2.F KRS Investment Organization



Source: Staff analysis.

Reporting

KRS staff report quarterly to the Investment Committee. Occasionally, staff make informational presentations, such as describing the advantages and disadvantages of investing in emerging markets.

The Kentucky Retirement Systems publishes an annual financial report showing all receipts, disbursements, assets, and liabilities, including a copy of the audit. The board may select an independent certified public accountant or the auditor of public accounts to perform the audit. A copy of the audit must be sent to the Legislative Research Commission (KRS 61.645(12)).

Internal and External Management

About one-third of total assets are managed internally. By category, KRS staff managed about 57 percent of total domestic equities and 33 percent of fixed-income investments in FY 2007. For the most part, internally managed funds attempt to mimic representative indices, a passive investment strategy. Internally managed funds outperformed externally managed domestic equity assets and underperformed externally managed fixed-income assets (Commonwealth. Kentucky Retirement. "Kentucky").

About one-third of total assets are managed internally, generally using passive funds.

Chapter 3

Kentucky Teachers' Retirement System

This chapter describes the Kentucky Teachers' Retirement System. It begins with a summary and continues by describing membership, funding, and governance. The chapter concludes with a description of KTRS's investment policies and practices.

Summary

The Kentucky Teachers' Retirement System was established in 1938 out of concerns that teachers could not afford to retire and school districts could not attract and retain educators (Commonwealth. Teachers'. "KTRS"). The primary purpose was to provide retirement benefits to educators and other public employee members of the retirement system. As of June 30, 2007, total membership was 119,100. Members do not participate in the federal Social Security system.

This chapter has five main points.

First, KTRS is governed by a board of nine trustees: seven elected and two ex officio. Board and committee members are not required to have investment experience. An investment committee of two trustees and the executive secretary oversees investments.

Second, net assets grew \$3.7 billion, or 31 percent, from FY 2002 to FY 2007. KTRS had \$15.6 billion in net assets at the end of fiscal year 2007.

Third, KTRS is changing its asset allocations by increasing its investments in international securities, private equity, and real estate, while decreasing domestic equity and short-term investments. Statute requires KTRS to set out its asset allocation policy in administrative regulation; thus, changing the policy requires changing administrative regulations.

Fourth, interest income and dividends plus gains in asset values accounted for between 45 percent and 68 percent of annual net plan additions from FY 2005 to FY 2007. Employer and member contributions were the next two largest components, with other revenues, including general fund appropriations of \$12 million in FY 2007 and \$14 million in FY 2008, representing the balance.

KTRS invests funds and manages benefits for 119,100 members. KTRS members do not participate in the federal Social Security system.

This chapter has five main points.

1. KTRS is governed by a ninemember board of trustees. Board and committee members are not required to have investment expertise. An investment committee of two trustees and the executive secretary oversees investments.

2. Net assets grew \$3.7 billion, or 31 percent, from FY 2002 to FY 2007.

3. KTRS is changing its asset allocations to include more investments in international securities, private equity, and real estate.

4. Investment income accounted for 45 percent to 68 percent of annual net plan additions between fiscal years 2005 and 2007.

5. KTRS's medical insurance fund has received and borrowed a total of \$624 million from its pension fund in order to maintain health care benefits. Fifth, KTRS's medical insurance fund has borrowed and received a total of \$624 million in funds designated for its pension fund in recent years. This was done so KTRS could continue paying annual insurance benefits without reducing benefit levels for members.

Membership

KTRS members include employees of school districts, five regional universities, the Kentucky Community and Technical College System, Department of Education agencies, and others. Kentucky Teachers' Retirement System membership includes employees from 198 employers including every school district, five regional universities, the Kentucky Community and Technical College System, Department of Education agencies, and others.¹ Per statute, eligible participants include certified employees and graduates of a 4-year college or university (KRS 161.220).

Figure 3.A shows the number of active, retired, and inactive KTRS members from FY 2002 through FY 2007.

Figure 3.A Total KTRS Membership by Member Classification, Fiscal Years 2002-2007



Source: Staff's compilation of information from Commonwealth. Kentucky Teachers'. *Comprehensive* FY 2002-FY 2007.

¹ The five regional universities in KTRS are Eastern Kentucky, Western Kentucky, Morehead, Murray State, and Kentucky State.

Between fiscal years 2002 and 2007, active membership rose 39 percent, and retired membership rose 18 percent. Most of the active membership increase was due to the addition of part-time and substitute teachers to the retirement system in FY 2002.

From FY 2003 to FY 2007, the ratio of active to retired members steadily declined.

Funding comes from four primary sources: employer contributions, member contributions, interest income and dividends, and net asset appreciation. Other revenue in recent years has included appropriations of \$12 million and \$14 million from the Kentucky general fund.

Employer and member contribution rates are set by statute.

Between FY 2002 and FY 2007, active membership increased 39 percent, from 54,200 to 75,100. Legislation enacted in 2002 allowed part-time and substitute teachers into KTRS, which mostly accounts for the increase. The number of retired members and their beneficiaries rose 18 percent, from 33,500 to 39,500. Inactive membership fell 26 percent, from 5,700 to 4,500. Total membership—the combination of active, retired, and inactive members—rose 27 percent, from 93,600 to 119,100.

Between FY 2002 and FY 2007, the ratio of active members to retiree and beneficiary members rose because part-time and substitute teachers were added to the active ranks. Considering only the period from FY 2003 to FY 2007 to account for this change, the active to retired ratio decreased from 2.1 to 1 to 1.9 to 1.

Funding

Pension and medical insurance benefits are provided by the retirement system to eligible members by separate pension and insurance funds. Funding comes from four primary sources: employer contributions, member contributions, interest income and dividends, and net asset appreciation. Other revenues included state appropriations of \$12 million in FY 2007 and \$14 million in FY 2008 to help shore up KTRS's insurance fund.

Employer and member contributions for both funds are based on salary. Statute establishes employer and member contribution rates. Employers and members each contribute an additional threequarters of 1 percent of salary to the insurance fund.

Statutes and Regulations

Statutory authority for KTRS is provided under KRS Chapter 161. KRS 161.430 covers investments, KRS 161.250 covers the composition and terms of the board, and KRS 161.340 covers officers and their liability.

Administrative regulations function in place of bylaws for KTRS. These are found at 107 KAR 1:175. Regulations pertain to the investment committee, investment counselors, and asset allocation policy.

Governance

KTRS is administered and managed by a board of trustees. Full power and responsibility to buy, sell, exchange, transfer, or otherwise dispose of investments rests with the board.

Trustees

The board consists of nine members: seven elected and two ex officio. The elected members are four teachers, two lay persons, and one retired teacher. Ex officio members are the chief state school officer and the state treasurer. Elected trustees receive a per diem when the board is in session. Elected board trustees serve 4-year terms.

The board hires an executive secretary under a contract of no more than 4 years. The executive secretary administers investment policies and procedures established by the board.

The board appoints an investment committee consisting of the executive secretary and two trustees. The investment committee makes recommendations to the board regarding employment of investment counselors and whether those contracts are renewed. Although there is no requirement that board members have investment expertise, the board is required by statute to employ experienced, competent investment counselors to advise it on all investment matters.

In general, no investment or disbursement of funds can be made unless authorized by the board. By administrative regulation, however, the board establishes investment guidelines under which staff and investment counselors may execute purchases and sales of investment instruments without prior board approval to ensure timely market transactions.

Fiduciary Responsibility

Trustees have a fiduciary responsibility to KTRS members. They must discharge their duties solely in the interests of members and retirees, following the "prudent person" standard. The standard states that the same care, skill, prudence, and diligence should be used as a prudent person with a similar capacity and in a similar environment.

The KTRS Board of Trustees includes seven elected and two ex officio members. Board members are not required to have investment expertise.

The board appoints an investment committee of two trustees plus the executive secretary.

KTRS's Board of Trustees has ultimate responsibility for investments and follows the "prudent person" standard.

Liability and Insurance

Trustees, the executive secretary, and other employees identified by the board are required to execute bonds for the faithful performance of their duties. KTRS carries a commercial crime bond of \$1 million on trustees and all employees. The board also purchases fiduciary liability coverage that includes legal defense coverage. Coverage includes breach of fiduciary duty committed by the KTRS Trust and any past, present, or future trustee or employee of the Trust (Harbin. Letter. April 4, 2008). Statute allows use of KTRS funds to insure against liability caused by an act or failure to act while performing assigned duties.

Assets

Figure 3.B shows net plan assets, which reflect contributions received, interest income, and asset appreciation less benefit payments and other expenses.

Between fiscal years 2002 and 2007, net plan assets grew from \$11.9 billion to \$15.6 billion, a 31 percent increase.

From FY 2002 to FY 2007, net plan assets rose from \$11.9 billion to \$15.6 billion, a 31 percent increase, which represents an average annual increase of 5.2 percent. Net plan assets increased by \$591 million in FY 2005, \$315 million in FY 2006, and \$1.65 billion in FY 2007. Of the two funds, KTRS's insurance fund has relatively minimal net assets.



Figure 3.B Net Plan Assets, KTRS Pension and Insurance, Fiscal Years 2002-2007

Source: Staff's compilation of information from Commonwealth. Kentucky Teachers'. *Comprehensive* FY 2002-FY 2007.

Summary of Changes in Net Plan Assets

In recent years, net plan assets increased primarily due to investment income, which is appreciation of investments plus interest income and dividends. Table 3.1 identifies the major components of changes in annual net plan assets by revenue and expense categories. Net investment appreciation and interest and dividends accounted for 53 percent in FY 2005, 45 percent in FY 2006, and 68 percent in FY 2007 of total annual additions. Annual net investment appreciation varied significantly from \$282 million in 2006 to \$1.6 billion in FY 2007, reflecting, in large part, an upswing in stock market returns. Interest and dividend income remained relatively stable at slightly more than \$400 million annually.

l able 3.1
Summary of KTRS's Changes in Net Plan Assets by Revenue and Expense Categories,
Fiscal Years 2005-2007, Pension and Insurance Funds (in thousands of dollars)

T II 31

	FY 2005		FY 2006		FY 2007	
		% of Sub-		% of Sub-		% of Sub-
	Total	total	Total	total	Total	total
Revenues						
Member contributions	\$298,601	17%	\$310,162	20%	\$322,788	11%
Employer contributions	\$467,369	27%	\$500,241	32%	\$553,147	18%
Net investment appreciation	\$512,314	29%	\$282,573	18%	\$1,617,338	54%
Interest and dividends	\$411,115	24%	\$415,938	27%	\$426,832	14%
Other income	\$33,122	3%	\$30,950	4%	\$30,344	3%
Subtotal	\$1,722,521	100%	\$1,539,864	100%	\$2,950,449	100%
Expenses						
Benefits	\$1,105,721	97%	\$1,200,612	96%	\$1,276,362	95%
Administrative	\$10,724	1%	\$11,464	1%	\$12,003	1%
Investment	\$4,670	2%	\$6,157	3%	\$8,668	4%
Subtotal	\$1,121,115	100%	\$1,218,233	100%	\$1,297,033	100%
Additions less Deductions	\$602,103		\$322,440		\$1,654,758	
Other changes	\$(10,985)		\$(6,724)		\$(4,493)	
Net plan asset change	\$591,118		\$315,716		\$1,650,265	

Note: The table does not include 403 (b) additions or deductions because it is an optional deferred compensation program. Subtotals may not add to 100% due to rounding.

Source: Staff's compilation of information from Commonwealth. Kentucky Teachers'. *Comprehensive* FY 2005-FY 2007.

Employer and member contribution amounts, the other primary funding components, steadily increased from FY 2005 to FY 2007. As a percentage of total additions, these contributions fluctuated due to large swings in investment appreciation. Benefits accounted for more than 95 percent of total expenses.

Health care expenses have been primarily funded by borrowing and redirecting \$624 million from KTRS's pension fund.

Total investment expenses more than doubled from FY 2002 to FY 2007, primarily due to increases in fees paid to two external fund managers for managing significantly higher dollar amounts.

Assets must be prudently diversified.

Asset allocation limits are broad investment parameters that must be written into regulation and reviewed by the Legislative Research Commission's Administrative Regulation Subcommittee. Benefits accounted for more than \$1 billion in pension and insurance expenses annually from FY 2005 to FY 2007. This represented more than 95 percent of total expenses.

Health care expenses are primarily funded by borrowing and redirecting funds from the pension fund. Over the past 10 years, KTRS, with legislative approval, has redirected \$624 million from its pension fund to its insurance fund.²

Investment expenses include management, consultant, and bank fees; but not professional consultant fees such as those paid to the actuary, auditor, or legal counsel.³ Between FY 2002 and FY 2007, total investment expenses rose from \$4.2 million to \$8.7 million, which represents an average annual increase of 17.4 percent. Fees to two equity managers (UBS and Wellington) doubled over a 2-year period, accounting for most of this increase. These increases were attributable to significant increases in the dollar amount of assets managed by these firms.

Asset Allocation Targets

Assets must be prudently diversified among various classes of investments. In determining asset allocation policy, the investment committee and the board must be mindful of the system's liquidity and its capability of meeting both short- and long-term obligations.

The Kentucky Teachers' Retirement System's asset allocation policy has allocation limits, allocation targets, and specific restrictions. Asset allocation limits are broad investment parameters adopted by the KTRS Board of Trustees and then written into regulation with review by the Legislative Research Commission's Administrative Regulation Subcommittee.

² Specifically, \$335 million was redirected from the 3.25 percent supplemental pension contribution, and an additional \$289 was borrowed directly from the pension fund.

 ³ In FY 2007, professional consultant fees were \$1.8 million. In FY 2006, they were \$2.4 million.

Table 3.2 lists asset allocation limits by investment type. They are typically expressed as maximums. For example, corporate debt obligations cannot exceed 35 percent of total system assets.

Type of Investment	Limit
Guaranteed by U.S. government	No limit
Corporate debt obligations	35% or less of system assets
Common stocks or preferred stocks	65% or less of system assets
General stock index portfolios	25% or less of system assets
Stocks of companies domiciled outside the U.S.	15% or less of system assets and counted toward 65% common/preferred stock limit
Real estate (including equity, lease agreements, and real estate investment trusts)	10% or less of system assets
Alternative investments (including private equity, venture capital, and timberland)	10% or less of system assets
Additional categories of investments	10% or less of system assets and with board approval

Table 3.2 KTRS Investment Category and Asset Allocation Limits

Sources: 102 KAR 1:175, Sec. 2, updated June 2007; Commonwealth. Teachers'. Annual 157.

Having established asset allocation limits and specific guidelines, the KTRS board sets allocation targets, but they are not enacted into administrative regulation. Asset allocation targets are more precise investment guideposts that attempt to balance return and risk in the current market by establishing how much money should be invested in particular asset classes.

Prior to FY 2008, KTRS established allocation ranges, but it now establishes specific allocation targets. Table 3.3 shows KTRS asset allocation ranges for FY 2007 and specific targets for fiscal years 2008 and 2009. KTRS did not identify specific allocation targets until FY 2008.

Table 3.3Target Asset Allocations for KTRS Pensionand Insurance Funds, Fiscal Years 2007, 2008, and 2009

Asset Class	FY 2007	FY 2008	FY 2009
Domestic equity	50-60%	56%	53%
International equity	1-10%	9%	11%
Fixed income	30-45%	28%	28%
Real estate	1-5%	4%	4%
Cash	3-10%	2%	2%

Sources: Staff's compilation of information from Commonwealth. Teachers'. *Annual* 191; Harbin. Letter. May 8, 2008.

Program Review and Investigations

Asset allocation targets for the FY 2008 to FY 2009 period are consistent with national trends. KTRS expects to redirect some assets from domestic equities to international equities. Fixedincome allocation targets remain unchanged at 28 percent but are lower than the FY 2007 target.

The Kentucky Teachers' Retirement System also considers statutory, regulatory, and investment guidelines when setting investment policies. These typically constrain, but in some cases promote, investment choice. They can be grouped into three categories: investments to avoid, preferred investments, and limitations on the amount of certain investments.

Prohibited investments include undeveloped land (unless development plans are imminent), bullion, foreign currency, stamps, rare coins, or other collectibles. Except for undeveloped land, with specific board approval, KTRS can invest up to 10 percent of its assets in the above areas as part of the 10 percent or less reserved for "additional category of investments" (102 KAR 1:175, Sec. 4).

In-state investments receive priority by the board whenever they are calculated to improve the industrial development and enhance the economic welfare of the Commonwealth (KRS 161.430(1)). The board, however, is not required to make in-state investments.

Prohibited investments include undeveloped land, bullion, foreign currency, stamps, rare coins, and other collectibles.

In-state investments are not required, although the board gives priority to investments calculated to improve the industrial development and economic welfare of the Commonwealth. The Kentucky Teachers' Retirement System also maintains specific fixed-income and equity investment guidelines. Tables 3.4 and 3.5 list them.

Category	Guideline
Securities of a single issuer (unless issuer is U.S. government or government- sponsored enterprise)	No more than 5% of the assets of the system may be invested
A fixed-income investment	Must be rated, when purchased, within the four highest credit classifications identified by one of the major rating services
Private placement debt investment	Must be rated, when purchased, within the four highest credit classifications identified by one of the major rating services
Fixed-income investment portfolio as a whole	Must maintain an average rating equal to at least the second-highest credit classification
Investments in mortgages or mortgage- backed securities	Must be first mortgages on property located in the U.S. unless guaranteed by the U.S. government
Debt obligations of Canadian government and Canadian-domiciled corporations	No more than 5% in aggregate of the assets of the system may be invested
Other foreign debt	Shall not be purchased unless approved by the board as an additional category of investment

Table 3.4
KTRS Fixed-income Investment Guidelines

Source: 102 KAR 1:175, Sec. 3.

Equity and Real Estate Investments	Guidelines
System's position in a single stock	Shall not exceed 2.5% of the system's assets; shall not exceed 5% of the outstanding stock for that company unless the investment is part of a venture capital program
Real estate purchase conducted on a triple net lease basis	Shall involve a company that at the initial agreement generates one of the three highest credit ratings by a national credit rating service
Real estate investment Source: 102 KAR 1:175, Sec. 4.	Shall be judged on its total return potential

Table 3.5KTRS Equity Investment Guidelines

Actual Asset Allocations

Actual asset allocations depend on several factors, including market conditions, cash flow, and investment expectations. As a result, actual asset allocations and target allocations usually do not match. Figure 3.C compares KTRS's actual asset allocations in FY 2000 and FY 2007.

Figure 3.C KTRS Actual Asset Allocations for Pension and Insurance Funds, Fiscal Years 2000 and 2007



Source: Staff's compilation of information from Commonwealth. Teachers'. *Comprehensive* FY 2000-FY 2007.

As of June 30, 2000, KTRS held less than 1 percent of its assets in international securities, but by June 30, 2007, that increased to 7 percent. The cash asset allocation has declined. KTRS does not have any subprime mortgage investments. There are two noteworthy trends in Figure 3.C. First, KTRS recently entered the international equity market. As of June 30, 2007, approximately 7 percent of total assets were invested in this market segment, compared to less than 1 percent in FY 2000. Second, cash assets declined from 6.8 percent in FY 2000 to 3.6 percent in FY 2007. Cash assets generally have lower rates of return than other investments. According to staff, KTRS does not have any investments in subprime mortgages or hedge funds.

Investments

KTRS invests pension and insurance assets jointly, even though benefits are maintained separately.

KTRS's investment goals and objectives are not clearly detailed

in its financial statements.

Goals and Objectives

Investment goals establish the broad principles on which investment policies and procedures are established. Investment objectives are specific measures of those goals. KTRS's investment goals were not obvious in its financial statements, and its objectives were not measurable.

In its *Annual Investment Report*: June 30, 2007, KTRS identified two "goals," which appear to be what are usually defined as objectives: 1) attain a 10-year annualized total return that exceeds the Consumer Price Index by 3.5 percentage points, and 2) attain a total return similar to a comparable major index for each asset class. Program Review staff could not find any mention of investment "goals" in KTRS's FY 2007 *Comprehensive Annual Financial Report.*

KTRS identifies two primary investment "concerns," which appear to be what are usually considered goals: 1) "preservation of assets in order to provide an adequate retirement plan for active and retired members"; and 2) invest "funds in order to promote the industrial development and economic welfare of Kentucky" (Commonwealth. Teachers'. *Annual* 153). It is unclear if "concern" and "goal" have the same meaning.

The Kentucky Teachers' Retirement System lists two investment objectives, but neither can be measured (Commonwealth. Teachers'. *Comprehensive* Fiscal Year 2007 46).⁴ The first objective is to invest funds in the sole interest of members and beneficiaries while allowing for reasonable administrative expenses. KTRS's second objective is that invested assets should provide a reasonable rate of total return, with major emphasis on protecting principal. Because "reasonable" is used in both objectives, and no definition was provided, success or failure cannot be determined. More precise objectives are needed.

Recommendation 3.1

The Kentucky Teachers' Retirement System should report specific investment goals and measurable investment objectives.

Recommendation 3.1 The Kentucky Teachers' Retirement System should report specific investment goals and measurable investment objectives.

⁴ Objectives are also maintained for specific portfolios. KTRS has had the same two investment objectives since at least FY 2004.

Organization and Operation

KTRS investments are managed by 6 professionals and 12 support staff.

KTRS's investments are managed by six professionals, four of whom hold the Chartered Financial Analyst designation. Investments are specifically divided into three groups: equities, fixed-income, and alternatives. An investment officer executes equity trades and works on monthly and quarterly performance calculations. A retirement investment specialist tracks KTRS portfolio performance. A 12-member support staff provides assistance (Harbin. Letter. May 8, 2008). Figure 3.D depicts KTRS's organizational chart for investments.



Figure 3.D

Support Staff (12)

Source: Staff's analyses of Harbin. Letter. May 8, 2008 and Commonwealth. Teachers'. "KTRS."

The Kentucky Teachers' Retirement System contracts with an investment consultant for the purpose of advising the investment committee on asset allocations and portfolio risk and return, and helping evaluate the effectiveness of investment counselors. Since at least 2002, KTRS has contracted with Becker, Burke Associates.

Investment counselors are external fund managers who must acknowledge in writing their fiduciary responsibilities to the retirement fund. They must also be registered under the Federal Investment Advisors Act of 1940, be a bank as defined by that Act, or be an insurance company qualified to perform investment

services under the laws of more than one state (KRS 161.430(3)(b))

Reporting

Investment counselors report their results at least quarterly. The investment committee annually reports to the board on the performance and service of each investment counselor.

An annual report, audit, and actuarial valuation must be published by the board each year. The annual report describes KTRS's operations and the past year's fiscal transactions and balance sheet. The audit evaluates and describes the retirement system's finances. It may be conducted by the auditor of public accounts or an independent certified public accountant. The actuarial valuation determines the retirement system's financial condition, including its unfunded liability, and projects future retirement costs and necessary contribution rates.

Internal and External Management

No single investment counselor is allowed to manage more than 40 percent of the funds of the retirement system. By statute, KTRS staff are limited to managing no more than 50 percent of its assets based on book value (KRS 161.430(1)).

As of June 30, 2007, asset management was about evenly split between internal and external managers. Based on market value, external managers oversaw 51 percent and internal managers 49 percent.

Fund management was further split along asset class lines. KTRS internal investment staff managed, for the most part, short-term investments, stock index funds, bond funds, and real estate equity. According to KTRS staff, bond funds have been actively managed. External fund managers generally have managed longer-term, global securities; specific stock funds such as small-, mid-, and large-capitalizations; and alternative investments such as international equity and private equity (Commonwealth. Teachers'. *Annual*). Five investment counselors managed 15 external funds. Counselors are required to follow KTRS's objectives, constraints, and administrative guidelines.

Investment counselors report results at least guarterly.

An individual investment counselor is limited to managing no more than 40 percent of KTRS's funds.

KTRS staff are limited to managing no more than 50 percent of assets.

KTRS staff generally manage short-term investments, stock index funds, bond funds, and real estate equity. To ensure assets remain appropriately allocated, monthly reports are developed by KTRS staff. To ensure assets remain appropriately allocated, monthly reports are developed by KTRS staff that compare market values by asset class against established allocation ranges and administrative regulation limits. Asset classes that approach these limits are managed so they continue to conform to regulations and policies (Harbin. Letter. May 8, 2008).

Chapter 4

Measuring Performance

Chapter 4 evaluates investment rates of return for KRS and KTRS.

The second objective of this report is to evaluate investment rates of return for the Kentucky Retirement Systems and the Kentucky Teachers' Retirement System. The chapter begins by evaluating actual investment performance compared with four measures approved by KRS and KTRS. Next, the chapter presents KRS's and KTRS's investment performance from a national perspective. Governance and investment factors that may affect investment rates of return follow.

KRS and KTRS Investment Performance

Summary

KRS and KTRS outperformed most internally established investment performance benchmarks. Table 4.1 summarizes four KRS and KTRS investment performance measures: fund benchmarks, asset class benchmarks, actuarially assumed investment rates of return, and system longterm objectives. Each measure was developed or approved by KRS or KTRS.

Table 4.1 Summary of Actual Investment Performance Compared With Performance Measures Approved by KRS and KTRS

Performance Measure	KRS	KTRS
Fund Benchmarks	Outperformed	Mixed
Asset Class Benchmarks	Mixed	Outperformed
Actuarially Assumed Rates	Mixed	Mixed
System Long-term Objectives	Outperformed	Outperformed
	•	

Source: Program Review staff's analysis.

KRS outperformed its fund benchmarks and system long-term objective over 1-, 3-, 5-, and 10-year periods but showed mixed results when evaluating performance against its asset class and actuarially assumed rate benchmarks.

KTRS outperformed its asset class benchmarks and system longterm objective, but actual investment rates of return did not exceed The rates of return for KRS's pension fund and KTRS's combined pension and insurance funds were lower than averages for comparable systems. KRS's insurance fund exceeded the averages for most investment periods.

Rate of return in isolation says nothing about the risk involved. There is no single standard for what the trade-off between risk and return should be. the 1-year fund benchmark or the 10-year average of the actuarially assumed rate.

Relative to comparable public employee retirement systems, actual rates of return for KRS's pension fund and KTRS's combined pension and insurance funds were lower over the most recent 1-, 3-, 5-, and 10-year periods. KRS's insurance fund exceeded the average over the most recent 1-, 3-, and 5-year periods but not the 10-year period.

Benchmarks Provide Context

According to Merriam-Webster's dictionary, benchmarks are "points of reference from which measurements may be made." In other words, benchmarks provide context. Investment rates of return have little meaning without benchmarks.

A 5 percent investment rate of return for a year may be a great result for a year in which financial markets were generally down. If markets gained significantly more than 15 percent, for example, a 15 percent return for a year may not be so great. Benchmarks therefore provide the context in which to evaluate rates of return relative to returns for similar financial market segments.

Looking at returns in isolation also says nothing about the risk involved and how returns go up and down over time. Typically, higher return investments necessitate more risk of losing money, at least in the short term. There is no single standard for what the trade-off between risk and return should be. The decision will be based on the judgment of decision makers, taking into consideration the needs of the retirement system.

Retirement systems use and report several benchmarks to evaluate investment performance. Four are considered below: a fund benchmark, an asset class benchmark, the actuarial rate, and the long-term objective.

A fund benchmark measures how well a retirement system performed by comparing performance to that of markets for the kinds of investments the system makes. The most straightforward fund benchmark is calculated by taking the benchmark indices used for each asset class within a fund and weighting them by each asset class's target allocation or proportionate share of the fund. Some systems create customized fund benchmarks that do not directly match their asset class benchmarks or asset class Retirement systems use and report several benchmarks to evaluate investment performance.

Fund benchmarks measure performance of an entire fund compared to how relevant investment markets are doing.

Asset class benchmarks measure the performance of particular groups of investments within a fund.

Actuarially assumed investment rates project funding needed to pay future liabilities.

Long-term objectives establish the importance of managing investments for the long run.

KRS has separate benchmarks for its pension and insurance funds. KTRS has one fund benchmark.

Over the past 1-, 3-, 5-, and 10year periods, KRS's pension fund and insurance fund investments outperformed their fund benchmarks. allocations. Fund benchmarks may be reported for the current year and for longer-term investment periods.

An asset class benchmark measures the performance of particular groups of investments, such as domestic equities, international equities, fixed income, or cash. It is usually an indicator of whether fund managers did as well as the market. The benchmark is almost always an existing index, such as the Wilshire 3000 or Standard and Poor's 500. A system can also calculate a customized asset class benchmark by using weighted percentages from multiple indices. Presumably, this would more accurately reflect the types of investments a retirement system held within each asset class.

An actuarial rate is developed by an actuary and projects the funding needed to pay for future liabilities and, thus, the contribution rates necessary to pay for those liabilities. If a system's investment returns were consistently less than the actuarial rate, then projections of future liabilities and contributions would have to be revised.

A long-term objective is typically set by a retirement system and may be different from the actuarial rate. It establishes the importance of managing investments for the long run. Often, the long-term objective is to equal or exceed the inflation rate, typically the Consumer Price Index, by 3 to 5 percentage points.

Fund and Asset Class Benchmarks

Both KRS and KTRS evaluate the performance of their investments by comparing actual rates of return with fund and asset class benchmarks. KRS has separate benchmarks for its pension and insurance funds. KTRS has one fund benchmark. Investment consultants guide the selection of these benchmarks and trustees approve them.

Kentucky Retirement Systems. Table 4.2 shows reported actual investment performance compared with benchmarks for KRS's pension fund, insurance fund, and asset classes within those funds. Downward arrows represent performance below the relevant benchmark. The pension fund represents about 84 percent of total net plan assets; the insurance fund about 16 percent.

Over the past 1-, 3-, 5-, and 10year periods, actual rates of return for the U.S. equity, international equity, fixed-income, and alternative equity asset classes equaled or exceeded their corresponding benchmark 16 out of 29 times. Over the past 1-, 3-, 5-, and 10-year periods, KRS's pension and insurance funds consistently outperformed their investment benchmarks by as much as 0.9 percentage points. For these periods, actual rates of return for the U.S. equity, international equity, fixed-income, and alternative equity asset classes equaled or exceeded their corresponding benchmarks 16 out of 29 times.

K	RS's Reported Act and Fund Compa	tual Rates of Retur ared to Benchmark	·	
	1 year	3 voor	5 year	

Table 4 2

	1 yea	ar	3 ye	ar	5 yea	ır	10 y	ear
		Bench		Bench		Bench		Bench
Pension Fund	Return	-mark	Return	-mark	Return	-mark	Return	-mark
U.S. Equity	19.3 ↓	19.9	12.4 ↓	12.5	11.3 ↓	11.7	8.1	8.0
International Equity	24.4 ↓	27.5	21.4 ↓	22.8	16.6 ↓	18.3	n/a	n/a
Fixed Income	5.7	5.5	4.1	3.9	5.2	5.1	6.3	6.2
Alternative Equity	15.6 ↓	16.8	21.5	10.9	18.6	11.6	16.5	7.3
Total Fund	15.3	14.9	11.4	10.5	10.4	10.1	8.1	8.0
Insurance Fund								
U.S. Equity	19.8 ↓	20.3	12.2	12.1	11.2	11.2	7.8	7.0
International Equity	24.2 ↓	27.5	22.0 ↓	22.8	17.8 ↓	18.2	n/a	n/a
Fixed Income	3.9 ↓	4.0	3.9	3.8	6.1	6.0	6.8	6.8
Alternative Equity	16.7 ↓	18.5	12.7	11.5	11.4	11.4	n/a	n/a
Total Fund	19.3	19.1	13.7	12.9	12.3	11.8	8.4	8.0

Note: A downward arrow represents performance below the relevant benchmark. Source: Commonwealth. Kentucky Retirement. *Comprehensive* FY 2007.

Actual reported returns for three of four pension asset classes and all four insurance asset classes for the most recent 1-year period were lower than their asset class benchmarks. This apparent inconsistency with both fund benchmarks being met is due to investment returns for cash assets not being reported. Investment returns for cash holdings exceeded benchmarks for both funds. As shown in Table 4.2, actual reported returns for three of four pension asset classes and all four insurance asset classes for the most recent 1-year period were lower than their asset class benchmarks. However, the pension fund and the insurance fund both exceeded their fund benchmarks. That KRS's 1-year actual returns exceeded the fund benchmarks while none or only one of the reported asset class benchmarks were met is counterintuitive and can be confusing.

The problem is that investment returns for cash assets are not reported in the *Comprehensive Annual Financial Report*. In FY 2007, investment returns for cash holdings exceeded their benchmarks for both funds, which means that the pension fund met two of five asset class benchmarks, and the insurance fund met one of five asset class benchmarks. Anyone who had to rely only on Recommendation 4.1 is that KRS should report investment returns and benchmarks for all classes of assets.

Recommendation 4.2 is that KRS should report more detailed information on how its fund benchmarks are calculated.

the reported information would not know this and would have to assume that the benchmarks for asset classes and funds were inconsistent.

Recommendation 4.1

Subject to reporting standards, the Kentucky Retirement Systems should report investment returns and benchmarks for all classes of assets.

Another issue in comparing fund and asset class benchmarks is that KRS uses a customized fund benchmark. Most of the indices used in the fund benchmark seem to be based on the indices used for asset classes weighted by the target allocations for each asset class during the previous fiscal year. For example, the pension fund benchmark weights international equities at 15 percent, the same as its FY 2006 target allocation, by using the MSCI EAFE index. However, at least one index used for the fund appears to be weighted differently than the target allocation, and one index was used for the fund but was not reported as being used for any asset class.

There is no problem with KRS choosing to use customized fund indices. However, it would be helpful if supporting information were provided to clarify how the fund benchmarks are calculated, including how fund indices differ from asset class indices and how the weighting by asset class allocation is done.

Recommendation 4.2

The Kentucky Retirement Systems should report more detailed information on how its fund benchmarks are calculated.

KTRS's actual fund returns exceeded the 3-, 5-, and 10-year benchmarks but fell short of the 1year benchmark. **Kentucky Teachers' Retirement System.** Table 4.3 shows KTRS actual investment rates of returns and corresponding benchmarks by asset class and fund. Downward arrows represent performance below the relevant benchmark.

 Table 4.3

 KTRS Actual Rates of Return by Asset Class and Fund Compared to Benchmarks (in percent)

Pension and	1 ye	ar	3 y	ear	5 y	ear	10 y	ear
Insurance	Return	Index	Return	Index	Return	Index	Return	Index
Total Equity	20.6	20.6	12.1	11.7	10.7	10.7	7.8	7.1
Total Bonds	6.3	5.7	4.2	3.7	4.7	4.4	6.2	6.0
Total Real Estate	8.2	See Note	9.7	See Note	9.6	See Note	9.3	See Note
Cash Equivalents	7.2	2.7	4.8	3.2	3.4	3.0	4.2	2.7
Total Fund	15.3 ↓	15.7	9.3	9.3	8.5	8.5	7.1	6.7

Note: A downward arrow represents performance below the relevant benchmark. According to KTRS staff, benchmarking real estate assets is difficult due to real estate's relative illiquidity. A nationally recognized commercial real estate index resembles KTRS's real estate earnings over the long run but is less applicable in the short run.

Sources: Commonwealth. Teachers'. Annual; Harbin. Letter. May 8, 2008.

Each of KTRS's asset classes met or exceeded their benchmarks over the 1-, 3-, 5-, and 10-year periods. For the fund, actual investment rates of return equaled or exceeded the benchmark rates over the 3-, 5-, and 10-year periods but not the 1-year period.

Recommendation 4.3 is that KTRS should report its fund benchmark, including a description of how that benchmark was calculated. Each of KTRS's asset classes met or exceeded their benchmarks over the 1-, 3-, 5-, and 10-year periods. Fund benchmarks were not reported in KTRS's *Comprehensive Annual Financial Report*, but KTRS staff provided these benchmarks to Program Review staff. For the fund, actual investment rates of return equaled or exceeded the benchmark rates over the 3-, 5-, and 10-year periods but not the 1-year period.

According to KTRS staff, fund benchmarks are based on actual asset allocations as of June 30. Because asset allocations regularly change throughout the year, calculating the fund benchmark according to end-of-year allocations likely means the fund benchmark will either exaggerate or understate the relative importance of each asset class. For example, international equities have recently represented a larger share of KTRS's assets at the end of the year than at the beginning. This explains how KTRS could have missed its fund benchmark in FY 2007 while meeting each asset class benchmark.

Recommendation 4.3

The Kentucky Teachers' Retirement System should report its fund benchmark, including a description of how that benchmark was calculated. Of 17 comparable systems, none met its benchmark while missing all its asset class benchmarks as the KRS insurance fund did in 2007. None failed to meet its fund benchmark while meeting or exceeding all its asset class benchmarks as KTRS did in 2007.

Comparable retirement systems differed in how they developed their benchmarks.

Most of the comparable systems used a single index per asset class to benchmark domestic equities, international equities, fixed income, and cash. None of the comparable systems used a customized benchmark for domestic equities, as KRS did in FY 2007.

KTRS reported one combined benchmark for its U.S. and international equity investments.

Comparable States. Program Review staff examined benchmarks for 17 comparable public employee or teacher retirement systems, which are listed in Appendix D, that reported annual fund and asset class benchmarks. Of those, there were no states in which the fund met its benchmark while missing all of its reported asset class benchmarks as the KRS insurance fund did in 2007. Only one system (Idaho Public Employees) met its fund benchmark while meeting only one reported asset class benchmark as the KRS pension fund did in 2007. No system failed to meet its fund benchmark while meeting or exceeding all of its asset class benchmarks as KTRS did in 2007.

For 14 systems, there was a clear explanation for how the fund benchmark was developed. Five systems reported using a customized fund benchmark, as does KRS. Nine systems reported fund benchmarks calculated solely based on their own actual or targeted asset class allocations and asset class indices, as does KTRS.

Most of the 17 comparable retirement systems reported using a single index each to benchmark domestic equities, international equities, fixed income, and cash. However, choice of index varied.¹ A few retirement systems developed customized benchmarks for their international equity and fixed-income asset classes. These benchmarks typically incorporated multiple indices. None of the systems reported using a customized benchmark for domestic equities, as KRS did in FY 2007.

KTRS benchmarks cash allocations according to the Consumer Price Index, which, most recently, was 2.7 percent. A review of comparable systems found six that reported a cash benchmark, but all used Treasury bill returns that ranged from 5.1 to 5.3 percent.

KTRS reported one combined benchmark for its U.S. and international equity investments. Of comparable systems, 16 out of 17 reported separate U.S. equity and international equity benchmarks. KTRS has been increasing its investments in international equities, so it would be helpful if returns and

¹ Among the 17 comparable systems, retirement systems referred to fund benchmarks by various names, including static, policy, composite, blended, and passive. For U.S. equities, each system used one of six indices or created a custom index. For international equities, systems used indices maintained by Morgan Stanley Capital International or created custom indices. For fixed income, systems used indices maintained by Lehman Brothers or created custom indices. All six of the systems reporting a benchmark for cash assets used U.S. Treasury bill returns.

benchmarks were reported separately for U.S. and international equities.

Recommendation 4.4

The Kentucky Teachers' Retirement System should report separate U.S. and international equity asset class returns and benchmarks.

Table 4.4 identifies benchmarks by asset class for KRS and KTRS. Benchmarks were generally similar to the average of comparable states. One exception was that KRS's fixed-income benchmarks for its pension and insurance funds were notably lower than the average of comparable systems.

Among the 17 comparable retirement systems, asset class benchmarks varied in the most recent year. The average domestic equity benchmark was 19.9 percent but ranged from 15.1 percent to 20.6 percent. International equity benchmarks varied even more. The average rate was 28 percent, but the lowest rate was 18.7 percent and the highest rate was 30.7 percent. Fixed-income securities averaged 6.2 percent and ranged from 4 percent to 6.8 percent.

Table 4.4Benchmarks by Asset Class for 17 Comparable Systems,
KRS, and KTRS, Fiscal Year 2007

17 Comparable	U.S.	International	Fixed
Systems	Equity	Equity	Income
Average	19.9%	28.0%	6.2%
Minimum	15.1%	18.7%	4.0%
Maximum	20.6%	30.7%	6.8%
KRS-Pension	19.9%	27.5%	5.5%
KRS-Insurance	20.3%	27.5%	4.0%
KTRS	20.6%*		6.0%

Note: * KTRS reports a combined U.S. and international equity benchmark. Sources: Staff analysis of comparable retirement systems' financial documents. Appendix F has the list of sources.

Actuarially Assumed Investment Rates of Return

Retirement system actuaries establish assumed investment rates of return to, among other things, project unfunded liabilities and establish actuarially required contribution rates. As a measure of retirement system performance, investment rates of return should meet or exceed actuarially assumed rates.

Recommendation 4.4 is that the system should report separate U.S. and international equity asset class returns and benchmarks.

KRS's and KTRS's investment returns exceeded the actuarially assumed rates over the past 1-, 3-, and 5-year periods, but fell slightly short of the 10-year rate. Table 4.5 shows the actuarially assumed rates of return for KRS and KTRS compared to the 1-, 3-, 5-, and 10-year investment returns. Rates of return for KRS's pension fund exceeded the actuarially assumed rate in the 1-, 3-, and 5-year periods but fell slightly short in the 10-year period. Similar results occurred with KTRS.

Table 4.5KRS and KTRS Pension Fund Actuarially AssumedRates of Return Compared to Investment Returns (Percent)

	K	RS	KTRS		
Period	Return	Actuary	Return	Actuary	
1-year	15.3	7.8	15.3	7.5	
3-year	11.4	7.9	9.3	7.5	
5-year	10.4	8.1	8.5	7.5	
10-year	8.1	8.2	7.1	7.6	

Note: Decimals are rounded to tenths. KRS's current rate is 7.75 percent. The 3-, 5-, and 10-year figures are averages of the rates of relevant years. Actuarial rates are different for insurance funds in accordance with governmental accounting standards. Those rates are not included here. Sources: Commonwealth. Kentucky Retirement. *Comprehensive* various years; Commonwealth. Teachers'. *Comprehensive* various years.

Kentucky Retirement Systems. The Kentucky Retirement KRS's actuarially assumed rate of return has declined from 8.25 Systems' actuary develops, and the KRS Board of Trustees adopts, percent in FY 2005 to 7.75 assumed rates about every 5 years. For the most recent pension percent in FY 2007. fund actuarial valuation (January 2006), KRS's actuary projected a 7.75 percent annual net investment rate of return for the pension funds. From 2002 to 2005, the projected rate of return was 8.25 percent. Compared with actuarially assumed rates of return, KRS's actual returns were at least 2 percentage points greater over the 1-, 3-, and 5-year periods. For the 10-year period, the rate of return was 8.1 percent, slightly less than the actuarial rate of 8.2 percent. KTRS actuarially assumed rate of Kentucky Teachers' Retirement System. Rates of return for return has been 7.5 percent since KTRS, which includes the pension and the insurance funds, FY 1999. exceeded the actuarial rates over the 1-, 3-, and 5-year periods. The 10-year rate of return of 7.1 percent fell short of the average 10-year actuarial rate of 7.55. KTRS's assumed rate of return has been 7.5 percent since FY 1999.

Comparable Systems. Of the 25 comparable public employee or teacher retirement systems examined in this report, current actuarially assumed investment rates of return ranged from 7.25 percent to 8.5 percent. These rates are listed by system in Appendix D. The most commonly reported rate was 8 percent, which is higher than KRS's rate of 7.75 percent and KTRS's rate of 7.5. Table 4.6 shows the number of retirement systems by actuarially assumed rate of return.

Actuarially Assumed Rate of Return	Number of Systems
7.25%	2
7.50%	4
7.75%	2
8.00%	11
8.25%	3
8.50%	3
Total	25

Table 4.6Current Actuarially Assumed Investment Rates of Return for
Comparable Public Employee or Teacher Retirement Systems

Sources: Staff's compilation of information from comparable retirement systems' financial documents. Appendix F has the list of sources.

System Long-term Objective

Kentucky Retirement Systems. Until FY 2007, KRS's long-term investment objective was to achieve a rate of return that exceeded the Consumer Price Index-Urban (CPI-U), a measure of inflation, by at least 4 percentage points over two market cycles, typically 6 to 10 years. From 1997 to 2007, the CPI-U plus 4 percentage points averaged 6.6 percent. KRS's rate of return averaged 8.9 percent for the pension fund and 8.6 percent for the insurance fund, so KRS met its long-term objective.

For FY 2007, KRS changed its long-term objective from the CPI-U to the actuarially assumed rate of return, which was 7.75 percent. The 10-year rate of return for KRS's pension fund was 8.1 percent, which is greater than 7.75 percent.

KRS's current long-term investment objective is to achieve a rate of return that exceeds its actuarially assumed rate of 7.75 percent over a 6- to 10-year period.
KTRS's long-term investment objective, which KTRS labels a "goal," is to attain a 10-year annualized total return that exceeds inflation by 3.5 percentage points.

Considering other retirement system rates of return can be useful but cannot explain on its own why differences occurred. **Kentucky Teachers' Retirement System.** According to KTRS's investment report, its long-term objective, which KTRS labels a "goal," is to attain a 10-year annualized total return that exceeds the Consumer Price Index by 3.5 percentage points.² For the 10-year period ending June 30, 2007, the CPI-U plus 3.5 percentage points totaled 6.1 percent. KTRS's 10-year return averaged 7.1 percent, meaning KTRS outperformed its long-term investment objective (Commonwealth. Teachers'. *Annual*). This includes the pension and medical insurance funds.

Comparable Systems. Long-run investment objectives were found in 11 of 25 comparable public employee or teacher retirement systems examined for this report. These objectives are listed by system in Appendix D. Most objectives were identified as inflation plus an additional 3 to 5.25 percentage points. One system's long-run objective was a flat percentage rate. Two systems used a flat percentage rate along with an inflation-based measure.

Investment Performance of Comparable Systems

In this section, the investment performance of comparable public employee and teacher retirement systems is examined. Although differences in investment policies, member demographics, funding, asset size, and asset allocations may contribute to rate of return differences, examining investment returns for a comparable group provides a context in which to consider KRS's and KTRS's actual returns.

² KTRS uses the term "goal," not "objective."

Fund Returns Relative to KRS

Table 4.7 identifies fund rates of return for 20 state employee retirement systems that had active membership of between 50,000 and 250,000 (U.S. Census Bureau. *State*).³ This range was selected because KRS's active membership as of June 30, 2007, was 148,000, placing it about in the middle.

Table 4.7Investment Rates of Return for State Public Employee Retirement SystemsWith Active Membership Between 50,000 and 250,000

	$\mathbf{D}_{\mathbf{r}}$				
	Rate of Return (%)				
Retirement System	1 year	3 year	5 year	10 year	
Arizona Retirement System	17.8	11.9	11.0	8.4	
Employees' Retirement System of Georgia	14.7	n/a	8.5	n/a	
Public Employee Retirement System of Idaho	20.0	14.3	12.8	9.0	
State Employees Retirement System of Illinois	17.1	12.6	10.8	n/a	
Indiana Public Employees' Retirement Fund	18.2	12.8	11.8	8.4	
Iowa Public Employees' Retirement System	16.3	12.9	11.6	9.0	
Kansas Public Employees Retirement System	18.0	14.1	12.3	8.8	
Louisiana State Employees' Retirement System	18.5	14.5	15.4	n/a	
Maine Public Employees Retirement System	16.2	11.8	11.4	7.7	
State Retirement and Pension System of Maryland	17.6	12.4	11.3	7.2	
Massachusetts State Employees' Retirement System	19.9	16.2	14.3	10.1	
Public Employees Retirement Association of Minnesota	18.5	14.0	12.0	8.5	
Missouri State Employees' Retirement System	18.1	14.2	13.3	9.2	
Public Employees' Retirement System of Nevada	15.0	11.0	10.0	7.9	
Public Employees Retirement System of New Mexico	18.1	13.2	11.7	9.8	
Oregon Public Employees Retirement System	18.6	15.6	13.4	n/a	
South Carolina Retirement Systems	13.4	8.6	8.8	7.0	
Tennessee Consolidated Retirement System	13.2	9.1	8.3	n/a	
Employees Retirement System of Texas	13.9	11.8	11.2	7.3	
Washington Public Employees' Retirement System	21.3	17.0	14.0	9.7	
Average	17.2	13.0	11.7	8.6	
KRS (Pension only)	15.3	11.4	10.4	8.1	
KRS (Insurance only)	19.3	13.7	12.5	8.4	

Notes: Some system investments are pooled with other in-state retirement funds. The 10-year rate of return for Indiana was calculated by staff based on annual data.

Sources: Various retirement system documents and staff's analysis. Appendix F has the list of sources.

³ Twenty-eight systems fell within this range, but eight were excluded due to old or missing data.

Public employee retirement systems generally experienced extraordinarily high returns in recent years due to increases in domestic and international stock market values.

Rates of return for KRS's pension fund were below average relative to 20 other systems but above average over the 1-, 3-, and 5year periods for its insurance fund.

Retirement systems that invested proportionately more assets in international equities generally outperformed other systems.

Relative to the average returns of nine other similarly sized retirement systems, KTRS investments did not perform as well. Retirement systems listed in Table 4.7 generally experienced extraordinarily high returns in recent years. This is evident by comparing the 1-year average return of 17.2 percent to the 10-year average return of 8.1 percent. The past 10-year period includes both the run-up and the collapse of the dot-com bubble and the post-9/11 recession.

Rates of return for KRS's pension fund were lower than the 20-state average for each time period. KRS's insurance fund returns were higher than the 20-state average for the 1-, 3-, and 5-year periods, but not the 10-year period.

Program Review staff's examination of 2007 asset allocations for these 20 retirement systems indicated that a relationship between risk and return existed. Relative to 1-year returns, retirement systems that invested proportionately more assets in international equities, a higher-risk investment, generally had higher rates of return than other systems.⁴

Fund Returns Relative to KTRS

Table 4.8 shows rates of return for 1-, 3-, 5-, and 10-year periods for state teacher retirement systems with 50,000 to 100,000 active members. That particular range was selected because KTRS membership falls in the middle with 75,000 active members.⁵ Thirty-two states operate teacher retirement systems (staff's analysis of U.S. Census Bureau data).

Average 1-year investment performance for the nine state teacher retirement systems was 18.5 percent. Exceptionally strong equity returns likely bumped rates above their historic average. For the 10-year period, the average return for these teacher retirement systems was 9.4 percent. Relative to the averages for these systems, KTRS's investment rates of return were lower for each time period.

⁴ Percent of assets allocated to international equities and 1-year rates of return had a 0.55 correlation value. A correlation value of 1 indicates that two factors move in unison; a correlation value of 0 indicates unrelated behavior.

⁵ Twelve retirement systems were originally considered, but three were excluded because the investment data were too old or could not be located.

	Rates of			
System	1 year	3 year	5 year	10 year
Arkansas Teachers Retirement System	19.1	14.0	12.1	n/a
Georgia Public School Employees Retirement System*	14.7	n/a	8.5	n/a
Indiana State Teachers' Retirement Fund	18.2	12.9	11.6	n/a
Louisiana Teachers' Retirement System	19.7	15.0	14.0	9.7
Massachusetts Teachers' Retirement System*	19.9	16.2	14.3	10.1
Minnesota Teachers' Retirement System*	18.5	14.0	12.0	8.5
Public School and Education Employee Retirement	16.6	11.8	10.5	8.2
Systems of Missouri				
Teachers' Retirement System of Oklahoma	18.5	12.8	12.8	10.4
Washington Teachers' Retirement System*	21.3	17.0	14.0	9.7
Average	18.5	14.2	12.2	9.4
KTRS	15.3	9.3	8.5	7.1

 Table 4.8

 Investment Rates of Return for State Teacher Retirement Systems With Active Membership Between 50,000 and 100,000

*Pooled with other in-state retirement funds.

Notes: The 10-year rate of return for Oklahoma was calculated by staff based on annual data.

Sources: Various retirement system documents and staff's analysis. Appendix F has the list of sources.

Asset Class Allocations and Returns Relative to KRS and KTRS

Allocations. Twenty-two comparable systems reported U.S. equity, international equity, and fixed-income allocations for FY 2007. Table 4.9 shows their range of asset class allocations compared to KRS and KTRS. Allocations ranged from 23 percent to 56 percent for U.S. equities, 0 percent to 30 percent for international equities, and 0 percent to 39 percent for fixed-income securities.

Table 4.9							
Asset Class Allocations for KRS, KTRS, and 22 Comparable Systems							

		U.S. Equities	International Equities	Fixed Income
Comparable	Average	40.6%	18.1%	24.1%
Systems	Minimum	23.3%	0.0%	0.0%
	Maximum	56.0%	29.8%	38.5%
	KRS-Pension	38.4%	18.4%	23.1%
	KRS-Insurance	55.5%	20.4%	0.0%
	KTRS	58.4%	6.7%	32.4%

Sources: Various retirement system documents and staff's analysis. Appendix F has the list of sources.

Relative to the average, KRS's pension fund allocated about 2 percentage points less to U.S. equities, while KRS's insurance fund allocated about 15 percentage points more and KTRS allocated about 18 percent more.

For international equities, KRS's pension fund allocated 0.3 percentage points more and KRS's insurance fund allocated 2.3 percentage points more than the average. KTRS allocated 11.4 percentage points less.

For fixed-income securities, KRS's pension fund allocated 1 percentage point less. KRS's insurance fund did not allocate any assets to fixed income. KTRS allocated 8.3 percentage points more than the average.

Returns. As noted previously, investment rates of return vary from one retirement system to another due, in some measure, to asset allocation differences. Some retirement systems had similar asset allocations but still had notable differences in rates of return. These differences were likely attributable to judgments of decision makers. Decision makers ultimately decide whether to manage assets internally or externally, how much risk to take within a particular asset class, and which investment managers to contract with. Each of these factors may positively or negatively impact rates of return.

Appendix E identifies domestic equity, international equity, and fixed-income asset allocations for each retirement system examined here. It also lists 1- and 5-year investment rates of return per asset class.

Governance and Investment Practices of Comparable Systems

While all public employee retirement systems have a similar goal of maintaining sufficient assets to pay benefits to members, the way retirement systems invest and oversee assets varies. In the following tables several important governance and investment characteristics specific to retirement systems are described. Program Review staff identified these characteristics by examining state statutes, administrative regulations, and financial statements for comparable state employee and teacher retirement systems.

Public employee retirement systems vary in how they invest and are governed. Retirement system boards include some combination of elected, appointed, and ex officio members. They have ultimate responsibility for investments. Comparable retirement system boards have between 4 and 18 members. KRS and KTRS have nine members each.

Most comparable retirement systems voluntarily create an investment oversight body.

Most comparable retirement systems require board or investment committee members to have some investment expertise or knowledge. Neither KRS nor KTRS has this requirement. Legislative Research Commission

Governing Bodies

Retirement system boards include some combination of elected, appointed, and ex officio members who have ultimate responsibility for investments. Comparable state government employee retirement systems, as shown in Table 4.10, have from 4 to 18 board members. Comparable teacher retirement systems, as shown in Table 4.11, have from 4 to 16 board members. KRS and KTRS have nine board members each.

Ten of the 20 public employee retirement systems do not have elected member representation. Four of nine teacher retirement systems do not have elected members. By comparison, KRS has five elected members and KTRS has seven.

Governance Requirements

Few of the comparable retirement systems are required by statute or regulation to operate an investment committee. However, most systems, including KRS and KTRS, voluntarily created some type of investment oversight body.

For most comparable retirement systems, state statutes or administrative regulations require board or investment committee members to have at least some investment expertise. Similar requirements do not exist for KRS or KTRS members.

Investment training for committee members is required by two of the comparable retirement systems. KRS and KTRS encourage, but do not require, training.

Investing Factors

Asset allocation limits cap the amount or percent of investments in particular assets. For example, a 40 percent domestic equity allocation limit means no more then 40 percent of a system's assets can be invested in domestic equities.

Statutory international investing restrictions may include a prohibition on investment in foreign securities; limitations on the percentage of assets that can be invested in foreign stocks, bonds, and other securities; or a requirement regarding the expertise of the investment manager.

In-state investing requirements generally mandate that a retirement system invest a defined percentage or amount of assets within the state in which it operates. Only two retirement systems examined here impose such a requirement. Language encouraging in-state investments "where prudent" is not considered a requirement for purposes of this report.

Table 4.10
Selected Governance and Investment Characteristics for State Public Employee
Retirement Systems With Active Membership Between 50,000 and 250,000

	Bo	oard N	Iembe	rs	Governa	nce Requi	irements	Inv	esting Fact	ors
Retirement System	Elected	Appointed	Ex Officio	Total	Investment committee	Investment expertise	Member training	Asset allocation limits	International investing restrictions	In-state investing requirement
AZ	0	9	0	9						
GA	-									
ID	1									
IL										
IN										
IA										
KS										
LA										
ME										
MD										
MA										
MN										
MO										
NV										
NM										
OR										
SC										
TN									5	
ΤX										
WA										
	Nu	mber	of Sys	tems	7	11	2	7	4	2
KRS	5	3	1	9						

Notes: Governance describes the body having final say on a system's investments and includes single-system retirement boards and state investment boards or commissions with investment responsibility for multiple systems. Required investment expertise may be on the governing board or investment committee. Source: Staff's analysis of various state statutes and administrative regulations.

61

Table 4.11
Selected Governance and Investment Characteristics for State Teacher Retirement
Systems With Active Membership Between 50,000 and 100,000

	Bo	oard N	lembe	ers	Governance Requirements			Inv	esting Fac	tors
Retirement System	Elected	Appointed	Ex Officio	Total	Investment committee	Investment expertise	Member training	Asset allocation limits	International investing restrictions	In-state investing requirement
AK	11	0	4	15						
GA										
IN										
LA										
MA										
MN										
МО										
OK				F						
WA				6						
Number of Systems		2	5	0	3	3	0			
KTRS	7	0	2	9						

Note: Required investment expertise may be on the governing board or investment committee. Oklahoma's regulatory asset allocations are guidelines and may be modified by the board. For Georgia, Massachusetts, Minnesota, and Washington, investments are pooled with their public employee systems' investments. Source: Staff's analysis of state statutes and administrative regulations.

Academic research indicates that retirement systems that invest proportionately more money in higher-risk investments typically achieve higher long-run returns than other systems.

Factors Related to Investment Performance

Public employee retirement systems earn different rates of return on their investment portfolios. As shown in Tables 4.7 and 4.8, some systems reported rates that exceeded 21 percent in FY 2007, others reported returns of less than 14 percent. Academic research has identified several factors that explain some of the difference. Most important is how retirement systems allocate assets. All things equal, retirement systems that invest proportionately more assets in higher-risk investments achieve higher long-run returns than other systems (Brinson).

Among retirement systems with similar asset allocations, asset size and in-state investing have consistently been identified as factors affecting rates of return (Munnell et al).

Asset size positively affects rates of return for two reasons. First, large systems generally operate more efficiently than small ones, in

part, because certain fixed operating expenses must be paid regardless of system size (Munnell and Sunden). Second, very large retirement systems may be able to influence markets through buying, selling, or persuading in a manner that improves rate of return. For example, the California state employee retirement system, which recently drew national attention for its decision to withhold votes for Eli Lily directors, has more than \$200 billion in assets (Graybow).

In-state investment requirements negatively impact rates of return. These requirements are imposed under the assumption that in-state investing can provide an appropriate rate of return while improving the local or state economy at the same time. With rare exceptions, these benefits do not occur. Worse, retirement systems reportedly end up receiving a lower rate of return than they would have received by investing elsewhere (Hsin; Schneider).

Many other governance and investment factors have also been examined, but conclusions have varied. Research has considered board size, percent of board members elected, board authority, and board transparency without consistent agreement (Romano; Munnell and Sunden; Coronado). The value of independent evaluations has also been debated (Useem and Mitchell; Munnell and Sunden). Studies considering the impact of tactical investing and socially responsible investing reached opposite conclusions (Useem and Hess; Nofsinger; Romano).

Some factors have no apparent effect on rate of return. These include prudent person requirements, external management, and reporting frequency (Mitchell and Hsin; Nofsinger; Useem and Mitchell).

Although governance structure may not directly affect rates of return, several studies have concluded that governance factors have an indirect effect. Governance factors may affect investment policy, which may affect investment rates of return (Useem and Mitchell).

In general, research identified few factors that might contribute to an optimal governance structure or investment policy for all public employee retirement systems. Only two investment factors—asset allocation and size—exhibited a consistent positive effect on returns. In-state investing requirements negatively affected rates of return. Governance factors appear to have no direct or consistent effect on rates of return, but no study has considered how board member investment expertise requirements might affect rates of

According to academic research,

- asset size positively affects rates of return because large systems generally operate more efficiently than small ones;
- in-state investment requirements negatively impact rates of return;
- board size, percent of board members elected, board authority, board transparency, independent evaluations, and tactical and social investing had mixed affects on rates of return; and
- governance factors may affect investment policy, which may affect investment rates of return.

return. These findings suggest high rates of returns over the long run can be achieved by allocating more assets to higher-risk investments, increasing asset size, and eliminating in-state investment requirements.

Works Cited

Ambachtsheer, Keith P. "The Economics of Pension Fund Management." *Financial Analysts Journal* 50.6 (1994): 21-31.

Brinson, Gary P., Brian D. Singer, and Gilbert L. Beehower. "Determinants of Portfolio Performance II: An Update." *Financial Analysts Journal* 47.3 (1991): 40-48.

Burnside, Robert M., Executive Director, Kentucky Retirement Systems. Letter to Rick Graycarek. May 2, 2008.

Commonwealth of Kentucky. Kentucky Retirement Systems. *Comprehensive Annual Financial Report*. For Fiscal Year Ended June 30, 2000, to Fiscal Year Ended June 30, 2007. Frankfort. 2000-2007.

---. Board of Trustees. "Statement of Bylaws and Committee Organization." April 5, 2007.

---. "Kentucky Retirement Systems Responses to Questions 1 Through 10 Regarding Retirement Systems Investment Study Program Review and Investigations Committee." April 4, 2008.

---. "Statement of Investment Policy-Pension Fund." Nov. 15, 2007.

---. "Statement of Investment Policy-Insurance Fund." Nov. 15, 2007.

---. KRS Investment Office. Year 2007-2008 Annual Five-Year Investment Plan. May 8, 2007.

---. Teachers' Retirement System of the State of Kentucky. *Annual Investment Report: June 30, 2007.* Frankfort. 2007.

---. *Comprehensive Annual Financial Report.* For Fiscal Year Ended June 30, 2000, to Fiscal Year Ended June 30, 2007. Frankfort. 2000-2007.

---. "KTRS Investment Overview." Feb. 8, 2008. Prepared by Gary L. Harbin.

Coronado, Julia L., Eric M. Engen, and Brian Knight. "Public Funds and Private Capital Markets." *National Tax Journal* 56.3 (2003): 579-594.

Eaton, Tim V., and John R. Nofsinger. "An Examination of ERR and Asset Allocation in Public Pension Plans." *Financial Accountability & Management* 17.2 (2001): 119-132.

Graybow, Martha. "Calpers wins backing for Eli Lilly bylaws proposal." Reuters. April 17, 2008. http://in.reuters.com/article/rbssFinancialServicesAndRealEstateNews/idINN1630166720080416 (accessed April 22, 2008).

Harbin, Gary L., Executive Secretary, Teachers' Retirement System of Kentucky. Letter to Rick Graycarek. April 4, 2008.

---. Letter to Rick Graycarek. May 8, 2008.

Hsin, Ping-Lung, and Olivia S. Mitchell. "Managing Public-Sector Pensions." *Public Policy Toward Pensions*. Ed. Sylvester J. Schieber and John B. Shoven. Cambridge, MA: MIT Press, 1997. 247-265.

Hustead, Edwin C. "Determining the Cost of Public Pension Plans." *Pensions in the Public Sector*. Ed. Olivia S. Mitchell and Edwin C. Hustead. Philadelphia: Univ. of Pennsylvania Press, 2001. 218-240. Mitchell, Olivia S., et al. "Developments in State and Local Pension Plans." *Pensions in the Public Sector*. Ed. Olivia S. Mitchell and Edwin C. Hustead. Philadelphia: Univ. of Pennsylvania Press, 2001. 11-39.

Mitchell, Olivia S., and Ping-Lung Hsin. "Public Pension Governance and Performance." *The Economics of Pensions*. Ed. Salvador Valdes-Prieto. Cambridge Univ. Press, 1999. 92-123.

Munnell, Alicia H., and Annika Sunden. "Investment Practices of State and Local Pension Funds." *Pensions in the Public Sector*. Ed. Olivia S. Mitchell and Edwin C. Hustead. Philadelphia: Univ. of Pennsylvania Press, 2001. 153-194.

Munnell, Alicia H., et al. "What Do We Know About the Universe of State and Local Plans?" *State and Local Pension Plans*. Center for Retirement Research at Boston College. March 2008.

Munnell, Alicia H., Steven A. Sass, and Mauricio Soto. "Yikes! How to Think About Risk?" *An Issue in Brief.* Center for Retirement Research at Boston College. Jan. 2005.

Mutual of America. "S&P 500." Oct. 2003. <http://www.mutualofamerica.com/articles/CapMan/October03/SandP500.htm> (accessed April 28, 2008).

Nofsinger, John R. "Why Targeted Investing Does Not Make Sense!" Financial Management 27.3 (1998): 87-96.

The Pew Center on the States. *Promises with a Price: Public Sector Retirement Benefits*. Dec. 18, 2007. <</ >
<www.pewcenteronthestates.org> (accessed Dec. 19, 2007).

Romano, Roberta. "Public Pension Fund Activism in Corporate Governance Reconsidered." *Columbia Law Review* 93.4 (1993): 795-853.

Schneider, Marguerite, and Fariborz Damanpour. "Determinants of Public Pension Plan Investment Return." *Public Management Review* 4.1 (2002): 551-573.

"South Carolina Makes Giant Leap." *Pensions and Investments*. Nov. 12, 2007. <PIonline.com> (accessed May 12, 2008.)

Standard & Poor's Index Services. "S&P 500 Monthly Returns, March 31, 2008." http://www2.standardandpoors.com/spf/xls/index/MONTHLY.xls (accessed May 30, 2008).

State of Wisconsin. Wisconsin Legislative Council. 2006 Comparative Study of Major Public Employee Retirement Systems. Dec. 2007. http://www.legis.state.wi.us/lc/publications/crs/2006_retirement.pdf (accessed Dec. 21, 2007).

Steffen, Karen. "State Employee Pension Plans." *Pensions in the Public Sector*. Ed. Olivia S. Mitchell and Edwin C. Hustead. Philadelphia: Univ. of Pennsylvania Press, 2001. 41-65.

United States. Board of Governors of the Federal Reserve System. "Flow of Funds Accounts of the United States." *Federal Reserve Statistical Release*. Dec. 6, 2007. http://www.federalreserve.gov/releases/z1/ (accessed May 29, 2008).

---. "Selected Interest Rates: Weekly." *Federal Reserve Statistical Release*. April 7, 2008. ">http://www.federalreserve.gov/releases/H15/> (accessed May 30, 2008).

---. Census Bureau. "Employee-Retirement Systems of State and Local Governments: 2002." 2002 Census of Governments 4.6 (Dec. 2004). http://www.census.gov/prod/2004pubs/gc024x6.pdf> (accessed May 16, 2008).

---. State & Local Government Employee-Retirement Systems Survey. Sep. 28, 2007. <http://www.census.gov/govs/www/retire.html> (accessed Dec. 21, 2007).

Program Review and Investigations

Useem, Michael, and David Hess. "Governance and Investments of Public Pensions." *Pensions in the Public Sector*. Ed. Olivia S. Mitchell and Edwin C. Hustead. Philadelphia: Univ. of Pennsylvania Press, 2001. 132-152.

Useem, Michael, and Olivia S. Mitchell. "Governance and Performance of Public Pension." *Social Science Quarterly* 81.2 (June 2000): 489-506.

Appendix A

Balancing the Equation

Understanding how investment performance fits into the overall funding of the retirement systems is essential. Investment income is only one part of the equation that makes up retirement funding. The basic equation is Employee Contributions + Employer Contributions + Investment Income = Pension Benefits + Expenses. If investment income decreases, then contributions must increase to pay for benefits and expenses. Further, if benefits and expenses rise or decrease, adjustments must be made on the other side of the equation to sustain the income needed to pay for benefits in the long run. Because pension benefits and member contributions are generally fixed, employer contributions and investment returns must balance the pension equation.

Another way to explain the fund is from a balance sheet perspective. If a plan is 100 percent funded, assets of the plan equal the benefits earned by and promised to active and retired members. When plan assets are less than promised benefits, the plan is less than 100 percent funded and has an unfunded actuarial accrued liability. The percentage funded depends on the difference between plan assets and promised benefits. This difference varies according to how much of the unfunded accrued liability (obligation) is met (paid) when it becomes due. Therefore, using the accounting equation, the overall management strategy should ensure a stable funding approach, depending on the pre-funding strategy as required by governmental accounting and reporting standards. The funding strategy affects investment strategy. A well-funded pension plan

- enhances investment performance;
- achieves the lowest employer contribution in the long term;
- provides additional assurance that promised benefits will be paid when due;
- provides full transparency of the actual cost of promised benefits; and
- complies with generally accepted accounting principles, based on accrual-based accounting and its ability to result in transparency of actual cost.

Principles of accrual-based accounting are the way public employee retirement systems are able to track long-term obligations and determine how they will pay for them when due. The accrual basis of accounting becomes the tool that managers rely on to know how strong the plan assets are, what the funding needs are, and what amount of investments can be allocated.

The Government Finance Officers Association's issued a 2005 policy statement that stated: The fundamental financial objective of a state or local government employee retirement system is to establish a funding policy, and receive contributions which, expressed as a percentage of active member payroll, will remain approximately level from generation-to-generation based on the plan's existing benefit package while fulfilling the long-term goal of fully funding member benefits. Embodied in this objective are the principles of accrual accounting, which require that the total cost of employee services be recognized in the period in which those services are rendered. The level contribution design of pension plan funding is intended to assign pension costs for the employee population to the appropriate fiscal periods.

Sources:

Government Finance Officers Association. "Funding of Public Employee Retirement Systems (1994 and 2005)." http://www.gfoa.org/downloads/corbafundingpublicemployeeretirementsyst.pdf>.

Kentucky Retirement Systems. *Member and Employer News*, Volume 25, Issue 2, May 2007. http://www.kyret.com/newsltrs/membernews200705.pdf>.

Kentucky Retirement Systems. *Comprehensive Annual Financial Report for fiscal year ended June 30, 2007.* http://www.kyret.com/cafr/cafr2007.pdf>.

Appendix B

Accounting Standards

Public employee retirement systems measure, recognize, and disclose future obligations for postemployment benefits, including pension and health care. The Governmental Accounting Standards Board (GASB) is an independent entity that establishes financial accounting and reporting standards for state and local governmental entities.

Pension Plan Standards

In 1994, GASB issued two government pension reporting statements to make pension information in governmental financial reports more understandable and useful. GASB Statement 25 established financial reporting standards for defined pension plans and note disclosures. GASB Statement 27 established standards for the measurement, recognition, and display of pension plan assets, expense, and related liabilities presented in financial reports of employers. All governmental employers, including school districts, must disclose actuarial and pension information in their financial statements.

Recently, GASB Statement 50 closely aligned the requirements of note disclosures and required supplementary information (RSI) for pensions with those for other postemployment benefit standards. Statement 50 improves the transparency and usefulness of financial reporting by pension plan trusts and employers by amending Statements 25 and 27, beginning after June 30, 2007. The following changes are required:

- Defined benefit pension plans must disclose actuarial methods and significant assumptions used in the most recent actuarial valuation in notes to financial statements, instead of RSI. The notes must disclose the funded status of the plan as of the most recent actuarial valuation date.
- If the aggregate actuarial cost method is used to determine actuarially required contributions, notes to financial statements must disclose the funded status of the plan, and a schedule of funding progress would be presented as RSI, using the entry age actuarial cost method.
- Notes to financial statements must include narrative disclosures of an informative nature regarding the actuarial measurement process. Further, the funded status in the note disclosures must be referenced and linked to the required schedule of funding progress in RSI.
- If applicable, notes to the financial statements must disclose legal or contractual maximum contribution rates.
- If an actuarial assumption is different for successive years, notes to financial statements must disclose the initial and ultimate rates.

Statement 50 also requires defined benefit pension plans and defined contribution plans to disclose in the notes to financial statements the methods and assumptions used to determine the fair value of investments, if based on other than quoted market prices.

Other Postemployment Benefit Standards

In 2004, GASB issued Statements 43 and 45, which required state and local governments to disclose the true costs of providing other postemployment benefits (OPEB). Prior to the standards established in these statements, it was not possible to analyze the financial position and long-term cost of healthcare and other postemployment benefits.

GASB Statement 43 established uniform financial reporting standards for postemployment plans other than pensions administered by trusts or equivalent arrangements. The standards are consistent with pension income standards. The main requirement is that the plan must report the actuarial cost and liabilities of the substantive OPEB plan based on the same actuarial cost methods allowed for pension benefits. The substantive plan may differ from written plan provisions based on whether benefits are discretionary or required by statutes. Assets are accumulated and benefits are paid as they come due in accordance with an agreement between employers and plan members and their beneficiaries, and in which

- employer contributions to the plan are irrevocable,
- plan assets are dedicated to providing benefits to their retirees and their beneficiaries in accordance with the terms of the plan, and
- plan assets are legally protected from creditors of the employer or the plan administrator.

GASB Statement 45 established accounting and financial reporting standards for any employer that provides postemployment benefits other than pensions by paying for all or part of the cost. The standard requires that the actuarial requirements be accounted for and reported in the employer's financial statements, unless the employer participates in a cost-sharing multiple-employer plan. In a cost-sharing plan, employers recognize annual OPEB expense for their contractually required contributions to the plan.

Employers are required to determine the actuarial costs of OPEB at least every 2 years for plans with 200 or more participants based on valuations performed in accordance with parameters. Standards require plans with fewer than 200 people to determine actuarial costs every 3 years. These valuations generally follow accepted actuarial practices established by the Actuarial Standards Board as follows:

- Future cash outlays are projected based on economic and demographic assumptions (as explained in the actuarial approach of financing).
- The cash outlays are discounted to their actuarial present value using a discount rate equal to an assumed long-term rate of return on investments. The actuarial assumption regarding investment returns should reflect the expected investment return on the OPEB assets rather than on the expected investment on employer assets.
- Once present value is determined, it is spread over a period approximating the anticipated length of service for an average employee, using one of six acceptable methods. GASB believes that governments should use the same method for both funding and financial reporting.
- The part of the actuarial present value allocated to a given year is the normal cost.
- The portion allocated to future years is called future normal costs.

Although Statements 43 and 45 are required for accounting and reporting, they do not mandate the funding of other postemployment benefits. Postemployment benefits are often financed in one of two ways. Some governments follow an actuarial approach, often referred to as the "contribution approach." Other governments follow a pay-as-you-go approach.

Employers applying the actuarial approach transfer an amount to the benefit plan that is expected to be sufficient, if currently invested, for future postemployment benefits. Actuaries determine the amount the employer should contribute based on appropriate assumptions. Assumptions about factors such as retirement ages, mortality schedules, inflation rates, investment returns, and demand for benefits are examples of assumptions used to estimate the cost of the benefits earned. This approach is commonly used to finance the costs of pension income.

Employers that apply the pay-as-you-go approach transfer only the resources required to satisfy the current benefits distributed or claimed in any given year. Many governments have been following this approach to finance OPEB.

GASB Statement 47 provides accounting and reporting standards for state and local governments that offer benefits such as early retirement incentives or severance benefits to employees that are involuntarily terminated.

Investment Standards

Governmental accounting and reporting standards require equity, debt, real estate, and other investments, excluding insurance contracts, to be reported at fair value. "Fair value" is the amount reasonably expected in a current sale between a willing buyer and a willing seller. Other investment requirements are very detailed and included in many GASB statements.

Sources:

For GASB Statements 25, 27, 43, and 45, information is from Codification of Governmental Accounting and Financial Reporting Standards as of June 30, 2006.

For GASB Statement 50, information is from J. Michael Inzina. "2008 GASB UPDATE." Kentucky Governmental Accounting and Auditing Conference. Executive West Hotel, Louisville. May 12-13, 2008.

Appendix C

Key Elements of Governmental Accounting Standards Board Statements 43 and 45 Relative to KRS and KTRS Insurance Funds

Key Elements of GASB	Insurance Fund	Insurance Fund
Statements 43 and 45	KRS	KTRS
Statements 45 and 45 Structure of Entity GASBS 43 applies to financial reports of plan sponsors of other postemployment benefits (OPEB) trust funds, and GASBS 45 applies to presentation requirements of governmental employers participating in the plan. GASB identifies types of postemployment benefit plans based on type of employer plan and how costs are determined. Reporting requirements vary based on the structure of the plan.	The KRS Insurance Fund is a cost-sharing, multiple-employer plan. It was established to provide hospital and medical insurance for all members receiving benefits from KERS, CERS, and SPRS. In 1988, medical coverage was added to the provisions of the inviolable contract (KRS 16.652, 61.692, and 78. 852). The fund pays a prescribed contribution for whole or partial payment of required premiums to purchase hospital and medical insurance, and for the spouse and dependents of retired hazardous-duty employees killed in the line of duty. The amount paid is a percentage based on years of service and is subject to annual changes based on CPI for all urban consumers. In addition, KRS began self- funding healthcare benefits for its Medicare-eligible retirees on Jan. 1, 2006, assuming the financial risk of providing healthcare benefits to its retirees.	The KTRS Medical Insurance Benefit Fund is a cost-sharing, multiple-employer plan. The plan provides post-retirement healthcare benefits to eligible members and dependents, as required by KRS 161.675. The fund has been self-insured since 1992. The dollar amount that KTRS pays toward coverage is directly related to the number of years of service credit at retirement. Coverage for spouses (under age 65), and dependent children of members are available at members' cost. Changes to the system may be made by the board of trustees, the General Assembly, and the Kentucky Department of Employee Insurance.
Implementation Dates	KRS implemented GASBS 43,	KTRS implemented GASBS
The criterion used to	required for fiscal year	43 for the fiscal year beginning
determine the	beginning after Dec. 15, 2005,	after Dec. 15, 2005, based on
implementation date for	and GASBS 45, required one	annual revenues of greater
GASBS 43 and GASBS 45	year later. Implementation	than \$100 million.
is based on total annual	requirements were based on	KTRS is in compliance with
revenues reported in the	annual revenues of greater than	implementation of the OPEB
first fiscal year ending after	\$100 million. KERS is the	standards.

Key Elements of GASB	Insurance Fund	Insurance Fund
Statements 43 and 45	KRS	KTRS
June 15, 1999. Plan revenues are based on the revenues of the largest participating employer for GASBS 45. Pre-funding and Funding Policy In order for an OPEB obligation to be considered	largest participating employer in the plan. KRS applied early implementation of GASBS 43 and GASBS 45. KRS is in compliance with implementation of the OPEB standards. KRS uses both contributions and investment income to fund the benefits of its Insurance	KTRS uses both contributions and short-term investment income to fund the post- retirement healthcare. KTRS
obligation to be considered funded, if only partially, the money must be set aside in a separate trust account. GASBS 43 refers to entities with a separate trust established in order to refund OPEB benefits under GASB regulations. KRS and KTRS initially had to determine whether to set aside funds and pay for the obligations with contributions and investment income or to use only contributions to fund the benefits.	Fund (OPEB plan). Management states that in prior years, the employers' required medical insurance contribution rate was increased annually by a percentage that would result in advance-funding the medical liability on an actuarially determined basis. KRS used the entry age normal cost method within a 20-year period measured from 1987. In Nov. 1992, the board of trustees adopted a fixed percentage contribution rate and suspended future increases under the current medical	retirement healthcare. KTRS uses a pay-as-you-go basis to finance its post-retirement healthcare (OPEB plan). To continue funding through 2008, the Commonwealth borrowed approximately \$289 million from the KTRS pension fund to meet healthcare obligations of KTRS plan members. Borrowing from the pension fund to sustain retiree medical insurance does impact actuarial soundness. KTRS prepares actuarial
 Other facts affecting funding: Short-term interest rates are lower than long-term rates. Pay-as-you-go financing has higher liabilities than fully funded OPEB plans. A blended rate may be used for employers using a blended funding approach. 	premium funding policy. In May 1996, the board adopted a policy to increase the insurance contribution rate by the amount needed to achieve the target rate for full entry age normal funding within 20 years. (See current actuarial cost method and UAL amortization.) In accordance with GASBS 43, lack of pre-funding the health insurance benefits in recent years has lead to a blended discount rate of 4.5%.	assumptions using both pre- funded and not pre-funded rates to show how each affects unfunded liability costs. KTRS's Schedule of Funding Progress reflects the change in discount rate of 4.5% because of lack of pre-funding. (See UAL amortization for both amounts.)

Key Elements of GASB Statements 43 and 45	Insurance Fund KRS	Insurance Fund KTRS
Asset-Valuation Methods Plan assets must be valued at a market-related value. This includes using actual market value or an asset- smoothing method that averages returns over a longer period, typically 3 to 5 years.	KRS uses a 5-year smoothed market related value for both non-hazardous and hazardous plans that averages returns.	KTRS uses a 5-year smoothed asset valuation market approach.
Actuarial Cost Method There are six actuarial cost methods that are acceptable under the parameters of GASBS 45. The cost method used should be the same method both for accounting and for funding calculations. The difference in the value between the actuarial accrued liability and the actuarial value of assets is known as the unfunded accrued liability (UAL), as defined in the section on amortization that follows.	KRS Uses an Entry Age Normal actuarial cost method. The actuarial present value of the projected benefits of each individual included in an actuarial valuation is allocated on a level basis over the earnings or service of the individual between entry age and assumed exit age.	KTRS uses a Projected Unit Credit actuarial cost method. The projected benefits of each individual included in an actuarial valuation are allocated by a consistent formula to valuation years.
Actuarial Assumptions The assumptions, referred to as parameters in GASBS 43 and 45, are assumptions such as investment return assumption (discount rate), healthcare cost trend, payroll growth, and other demographics. Actuarially determined amounts are subject to continual revision as actual results are compared with past expectations and new estimates are made about the future.	 KRS uses the following actuarial assumptions: Medical trend assumption of 12% - 5% (ultimate trend) for both nonhazardous and hazardous Year of ultimate trend is 2015. Medical trend assumption decreases by 1% each year up to 2015. Investment return assumption of 4.5% for nonhazardous and 7.75% for hazardous; includes inflation rate of 3.5% for nonhazardous and hazardous. 	 KTRS uses the following actuarial assumptions: Investment rate of return: 4.5%; includes inflation rate of 4% Projected salary increases: 4% Healthcare cost trend: 12% Ultimate healthcare trend: 5% Year of ultimate pre-Medicare trend rate is 2015. Medical trend decreases by 1% until ultimate percentage is reached in 2015

Key Elements of GASB	Insurance Fund	Insurance Fund
Statements 43 and 45	KRS	KTRS
Employers are required to determine the actuarial costs of OPEB based on evaluations performed in accordance with parameters established by GASB at least every 2 years for plans with 200 or more participants.	KKS	
Amortization of Any Unfunded Accrued Liability (UAL) Three elements are decided when determining how unfunded accrued liability	The amortization of the plan's UAL conforms to standards in accordance with the actuarial cost method of entry-age normal. The UAL is amortized over a 30-year period using a	KTRS amortization period conforms to accounting standards. KTRS uses level percent pay, open to amortize UAL over a
 is amortized. An employer must: use an amortization period of up to 30 years to meet the GASBS 45 compliance standard, decide whether the period is open or closed, and decide whether the payment will be flat dollar or will grow with total payroll. 	 level percent closed to covered payroll. UAL is determined separately for all members of the multi-cost sharing plan. As of June 30, 2007, the UAL amounts for KRS insurance fund were \$4,833.4 million for KERS (non-hazardous and hazardous), \$3,507.2 million for CERS (non-hazardous and hazardous), \$317.5 million for SPRS (hazardous). 	30 year period. UAL was \$5,787.9 million (not pre-funded discount) for valuation year of June 30, 2007, an increase from 2006. KTRS also determines a pre- funded discount rate of 7.5% that results in a UAL of \$3,514.2 million.

Sources:

Codification of Government Accounting and Reporting Standards (sections relating to Statements 43 and 45).

Kentucky Retirement Systems. Audited Financial Statements, June 30, 2007.

Kentucky Teachers' Retirement System. Annual Comprehensive Financial Statements, June 30, 2007.

"Kentucky Retirement Systems" and "Kentucky Teachers' Retirement System." Presentations to the Governor's Blue Ribbon Commission on Public Employees Retirement Systems. Frankfort. May 14, 2007.

Appendix D

Actuarial Assumptions and Benchmarks for Comparable Public Employee and Teacher Retirement Systems

This appendix details actuarial investment rate of return assumptions, long-term investment objectives, and policy objectives for those public employee retirement systems considered in Chapter 4. Retirement systems were selected on the basis of active membership relative to Kentucky Retirement Systems or Kentucky Teachers' Retirement System. The Georgia, Massachusetts, Minnesota, and Washington systems combine state employees and teachers and are covered in this appendix in the first table only.

Actuarial	Long-term Investment Objective	Fund Benchmark:	Fund Benchmark: More Than 1 Year	
	Objective	1 1 cai	1 I Cai	
0.00				
7 50				
7.50				
0.00	Tu Clatin u I			
8.00		N		
0.50	4./5%-5.25%	1		
8.50				
7.25				
7.50	Inflation + 3%			
8.00				
8.25	9.15% &			
	Inflation + 4%			
7.75				
7 75	Inflation $+ 3\%$			
8 25				
0.20			*	
8 50	Inflation +			
0.50	3%-5%		v	
	Assumption (%) 8.00 7.50 8.00 8.00 7.50 8.50 7.25 7.50 8.00	Actuarial Assumption (%) Investment Objective 8.00	Actuarial Assumption (%)Investment ObjectiveBenchmark: 1 Year 8.00 1 Year 7.50 1 flation + 4.75% - 5.25% 8.00 Inflation + 4.75% - 5.25% 7.25 $$ 7.25 $\sqrt{$ 7.50 Inflation + 3% 8.00 $\sqrt{$ 8.25 9.15% & Inflation + 4% 7.75 Inflation + 3% 7.75 Inflation + 3% 8.25 9.15% & Inflation + 4% 7.75 Inflation + 3% 8.25 $\sqrt{$ 8.25 $$ 8.25 $$ 8.25 $$ 8.25 $$ 8.25 $$ 8.25 $$ 8.25 $$ 8.50 Inflation + 3% 8.50 Inflation +	

Public Employees

Table continued on next page.

Public Employees continued

System	Actuarial Assumption (%)	Long-term Investment Objective	Fund Benchmark: 1 Year	Fund Benchmark: More Than 1 Year	
Missouri State Employees' Retirement System	8.50	Inflation + 5%			
Nevada Public Employees Retirement Board	8.00	Inflation + 4.5%			
New Mexico Public Employees Retirement	8.00				
Oregon Public Employees Retirement System	8.00				
South Carolina Retirement System	7.25	8.33%			
Tennessee Consolidated Retirement System	7.50				
Texas Employees Retirement System	8.00				
Washington Public Employees Retirement System	8.00				

Source: Staff's analysis of financial statements for other public employee retirement systems. Appendix F has the list of sources.

System	Actuarial Assumption (%)	Long-term Investment Objective	Fund Benchmark: 1 Year	Fund Benchmark: More Than 1 Year	
Arkansas Teachers	8.00				
Retirement System					
Indiana Teachers'	7.50				
Retirement Fund					
Louisiana Teachers'	8.25	Inflation + 3.9%			
Retirement System					
Missouri Public School	ouri Public School 8.00				
Retirement Board		Inflation $+4.5\%$			
Oklahoma Teachers'	8.00	Inflation + 3%		\checkmark	
Retirement System					

Source: Staff's analysis of financial statements for other public employee retirement systems. Appendix F has the list of sources.

Teachers

Appendix E

and Comparable 1 ubite Employee and 1 eacher Kethement Systems									
	FY 2007 Allocations (%)			1-year Returns (%)			5-year Returns (%)		
	Inter-			Inter-			Inter-		
	U.S.	national	Fixed	U.S.	national	Fixed	U.S.	national	Fixed
State	Equities	Equities	Income	Equities	Equities	Income		Equities	Income
AZ	56.0	20.0	23.0	19.8	27.9	6.1	11.7	17.2	4.6
ID	36.7	15.3	24.2	19.4	34.5	5.7	12.1	20.8	5.4
IL	52.0	11.0	20.0	19.9	28.8	6.9	11.8	17.9	4.7
IN	48.0	17.6	21.9	18.8	33.3	5.5	12.4	19.8	4.9
IN(T)	45.7	21.2	20.8	19.8	30.8	7.2	11.7	17.4	5.6
IA	29.7	15.6	32.5	19.4	27.7	6.5	12.0	18.8	5.4
KS	33.0	20.0	17.0	19.2	30.8	6.7	11.7	18.4	6.5
LA	39.3	24.9	20.3	19.3	32.7	7.0	12.0	21.4	7.3
LA(T)	37.1	22.5	16.7	15.8	30.2	NA	10.3	30.5	NA
ME	48.0	20.0	0	19.9	29.8		11.6	19.1	
MD	41.4	13.3	28.0	19.8	29.6	6.1	11.5	17.5	5.3
MA	25.1	21.0	15.4	20.8	26.5	4.9	12.2	17.8	5.2
MN	49.6	16.3	22.1	19.7	30.3	6.3	11.3	18.8	5.0
MO(T)	37.3	22.0	25.0	18.8	31.8	6.3	11.6	18.1	4.8
NV	45.1	10.5	33.5	20.2	27.0	NA	10.7	17.2	NA
NM	40.8	29.8	27.1	19.2	29.7	6.6	11.6	19.0	5.0
OK(T)	51.3	16.6	27.2	21.8	27.7	6.7		18.0	4.9
OR	28.6	21.6	26.8	19.7	30.0	7.0	12.1	20.3	6.6
SC	46.7	0	31.1	18.7		7.0	10.8		5.5
TN	33.2	15.5	38.5	20.6	27.2	NA	11.1	18.7	NA
ΤX	44.6	18.6	36.1	15.0	20.0	5.7	12.2	18.6	5.5
WA	23.3	23.8	23.5	20.3	30.7	6.6	12.0	19.5	5.7
Highest	56.0	29.8	38.5	21.8	34.5	7.2	14.2	30.5	7.3
Lowest	23.3	0.0	0.0	15.0	20.0	4.9	10.3	17.2	4.6
Average	40.6	18.1	24.1	19.4	29.4	6.4	11.8	19.3	5.4
KRS-									
Pension	38.4	18.4	23.1	19.3	24.4	5.7	11.3	16.6	5.2
KRS-									
Ins.	55.5	20.4		19.8	24.2		11.7	17.8	
KTRS*	58.4	6.7	32.4	20.6*		6.3	12.1*		4.7

Asset Allocations and 1-year and 5-year Investment Returns for KRS, KTRS, and Comparable Public Employee and Teacher Retirement Systems

Note: The table includes information for comparable states for which 1-year and 5-year investment returns were available. Teacher pension systems are denoted by (T); other systems are public employee systems or combined public employee/teacher systems. No overall returns for fixed-income investments were available for three states ("NA" in the table) because returns were reported separately for domestic and international fixed-income investments.

*In its financial reports, KTRS does not separate returns for U.S. and international equities. In this table, the returns are shown in the columns for U.S. equities.

Sources: Staff compilation of data from Commonwealth. Kentucky Retirement. *Comprehensive* FY 2007; Commonwealth. Teachers'. *Comprehensive* FY 2007; and documents for other systems. Appendix F has the list of sources for comparable systems.

Appendix F

Sources of Data for Comparable Public Employee and Teacher Retirement Systems

Sources of information for the 25 comparable systems analyzed in this report are listed below in alphabetical order by state.

Arizona State Retirement System. *Comprehensive Annual Financial Report for Fiscal Year Ended June 30, 2007*. Phoenix: 2007. http://www.azasrs.gov/web/pdf/financials/2007_54th_Annual_Financial_Report.pdf (accessed May 7, 2008).

Mercer Investment Consulting. "Total Fund Review-Second Quarter Board Meeting Arizona State Retirement System," meeting dates: Aug. 18, 2005; Aug. 18, 2006; Aug. 19, 2007. Los Angeles: 2005, 2006, 2007. http://www.azasrs.gov/web/Investments.do (accessed April 25, 2008).

Arkansas Teacher Retirement System. 2007 Comprehensive Annual Financial Report. Little Rock: 2007. Little Rock: 2007. http://artrs.gov/Forms/ATRS_10008_07_Annual_Report%202.pdf> (accessed May 2, 2008).

Employees' Retirement System of Georgia. 2007 Annual Report. Atlanta: 2007. http://www.ersga.org/forms.htm> (accessed April 25, 2008).

Public Employee Retirement System of Idaho. *Comprehensive Annual Financial Report As and for Fiscal Year Ended June 30, 2007*. Boise: 2007. http://www.persi.idaho.gov/html/generalinformation/Annual_Reports/annual_report.htm (accessed April 25, 2008).

State Employees' Retirement System of Illinois. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007.* Springfield: 2007. http://www.state.il.us/srs/SERS/annreports_sers.htm> (accessed May 2, 2008).

Indiana Public Employees' Retirement Fund. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007.* Indianapolis: 2007. http://www.in.gov/perf/files/investment_final010308.pdf> (accessed May 2, 2008).

Indiana State Teachers' Retirement Fund. 2007 Fiscal Operations Report. Indianapolis: 2007. http://www.in.gov/trf/files/2007_Fiscal_Operations_Report.pdf> (accessed May 8, 2008).

Iowa Public Employees' Retirement System. *Comprehensive Annual Financial Report the Year Ended June 30, 2007.* Des Moines: 2007.

http://www.ipers.org/publications/misc/pdf/financial/cafr/cafr.pdf (accessed May 2, 2008).

Kansas Public Employees Retirement System. *Comprehensive Annual Financial Report Fiscal Year Ended June 30, 2007.* Topeka: 2007. http://www.kpers.org/annualreport2007.pdf (accessed May 2, 2008).

---. "KPERS 10-Year Investment Return Rates." http://www.kpers.org/reportsinvchart3.pdf> (accessed May 2, 2008).

Louisiana State Employees' Retirement System. *Comprehensive Annual Financial Report Fiscal Year Ended June 30, 2007.* Baton Rouge: 2007.

<http://www.lasers.louisiana.gov/PDFs/Publications_and_Reports/Fiscal_Documents/Comprehe nsive_Financial_Reports/Comprehensive%20Financial%20Reports_07.pdf> (accessed May 5, 2008).

Teachers Retirement System of Louisiana. 2007 Comprehensive Annual Financial Report. Baton Rouge: 2007. http://trsl.org/ezedit/pdfs/cafr_07.pdf> (accessed May 5, 2008).

Maine Public Employees Retirement System. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007*. Augusta: 2007. http://www.msrs.org/pdfs/publications/07cafr.pdf> (accessed May 5, 2008).

State Retirement and Pension System of Maryland. *Comprehensive Annual Financial Report for the Year Ended June 30, 2007.* Baltimore: 2007. http://www.sra.state.md.us/annualreports/2007/CAFR2007.pdf> (accessed May 5, 2008).

Massachusetts Pension Reserves Investment Management Board. *Pension Reserves Investment Trust Fund Comprehensive Annual Financial Report Fiscal Year Ended June 30, 2007.* Boston: 2007. http://www.mapension.com/Publications/Annual/PRITCAFR2007.pdf (accessed May 5, 2008).

Minnesota State Board of Investment. Public Employees Retirement Association of Minnesota. *Comprehensive Annual Financial Report Fiscal Year Ended June 30, 2007.* St. Paul: 2007. http://www.msrs.state.mn.us/pdf/2007CAFR.pdf (accessed May 5, 2008).

---. 2007 Annual Report. St. Paul: 2007. http://www.sbi.state.mn.us/2007%20Annual%20Report.pdf> (accessed May 5, 2008).

Missouri State Employees' Retirement System. *Comprehensive Annual Financial Report Year Ended June 30, 2007.* Jefferson City: 2007. http://www.mosers.org/about/annu_report.asp (accessed May 6, 2008).

Public School and Education Employee Retirement Systems of Missouri. *Comprehensive Annual Financial Report for the fiscal year ended June 30, 2007.* Jefferson City: 2007. http://www.psrs-peers.org/Investments/AnnualReport.htm> (accessed May 9, 2008).

Public Employees' Retirement Board. Public Employees' Retirement System of Nevada. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007.* Carson City: 2007.

Public Employees Retirement System of New Mexico. "Complete Investments." http://www.pera.state.nm.us/pdf/FY07Q4InvestmentReport.pdf> (accessed May 6, 2008).

---. *Comprehensive Annual Financial Report Year Ended June 30, 2007.* Santa Fe: 2007. http://www.pera.state.nm.us/pdf/2007CAFRonline.pdf> (accessed May 6, 2008).

Teachers' Retirement System of Oklahoma. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007*. Oklahoma City: 2007. <http://www.ok.gov/TRS/documents/2007%20Annual%20Financial%20Report.pdf> (accessed May 9, 2008).

---. "Historical Investment Performance." http://www.ok.gov/TRS/documents/Historical%20Investment%20Performance%202007.pdf (accessed May 5, 2008).

Public Employees Retirement Board. Oregon Public Employees Retirement System. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007*. Salem: 2007. http://www.oregon.gov/PERS/docs/financial_reports/2007CAFR.pdf> (accessed May 6, 2008).

State Budget and Control Board. South Carolina Retirement Systems. *Comprehensive Annual Financial Report Fiscal Year Ended June 30, 2007*. Columbia: 2007. http://www.retirement.sc.gov/publications/cafr2007.pdf> (accessed May 7, 2008).

Treasury Department. Tennessee Consolidated Retirement System. *Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007*. Nashville: 2007. http://www.teasury.state.tn.us/tcrs/TCRS-AnnualReport.pdf (accessed May 7, 2008).

Employees Retirement System of Texas. *Comprehensive Annual Financial Report Fiscal Year Ended August 31, 2007.* Austin: 2007. http://www.ers.state.tx.us/about/financial reports/cafr 2007.aspx> (accessed May 7, 2008).

Department of Retirement Systems. *Funds of the State of Washington Comprehensive Annual Financial Report for the Fiscal Year Ended June 30, 2007.* Tumwater: 2007. ">http://www.drs.wa.gov/Administration/AnnualReport/CAFR/> (accessed May 7, 2008).

Washington State Investment Board. *Twenty-Sixth Annual Report 2007*. Olympia: 2007. http://www.sib.wa.gov/financial/pdfs/annual/ar07.pdf> (accessed May 7, 2008).